

# **Research progress of magnetorheological polishing technology:** a review

Ming-Ming  $Lu^1 \cdot Ya$ -Kun  $Yang^1 \cdot Jie$ -Qiong  $Lin^{1}$  · Yong-Sheng  $Du^1 \cdot Xiao$ -Qin Zhou<sup>2</sup>

Received: 25 April 2023 / Revised: 16 October 2023 / Accepted: 21 February 2024 © The Author(s) 2024

**Abstract** As an essential link in ultra-precision machining technology, various new surface polishing technologies and processes have always attracted continuous in-depth research and exploration by researchers. As a new research direction of ultra-precision machining technology, magnetorheological polishing technology has become an important part. The polishing materials and magnetorheological fluids involved in the process of magnetorheological polishing are reviewed. The polishing principle, equipment development, theoretical research and process research of magnetorheological polishing technologies, such as the wheel-type, cluster-type, ball-type, disc-type and other types, derived from the magnetorheological polishing process, are reviewed. The above magnetorheological polishing technologies are analyzed and compared from the perspective of processing accuracy, processing efficiency and application range. The curvature adaptive magnetorheological polishing technology with a circulatory system is proposed to achieve high efficiency and high-quality polishing.

**Keywords** Ultra-precision machining · Magnetorheological polishing · Research progress · Curvature adaptive · Development trend

Jie-Qiong Lin linjieqiong@ccut.edu.cn

<sup>1</sup> School of Mechanical and Electrical Engineering, Changchun University of Technology, Changchun 130012, People's Republic of China

<sup>2</sup> School of Mechanical and Aerospace Engineering, Jilin University, Changchun 130012, People's Republic of China

#### **1** Introduction

A new round of scientific and technological revolution and industrial change is breeding and emerging, and global scientific and technological innovation is showing new developments. The manufacturing sector is brewing a breakthrough in traditional processing methods to enhance the development of modern science and technology [1]. The significant instruments supported by ultra-precision machining technology, such as C919 aircraft, Jiao long aircraft and horizontal dual five-axis mirror milling machine, have played a decisive role in promoting the process of high-quality development of the equipment manufacturing industry [2]. The new functional materials supported by ultra-precision processing technology, such as semiconductor, optical and ceramic materials, have laid a solid foundation for developing aerospace, electronic information and civil industries. As an important supporting technology for the technological revolution and industrial change, ultra-precision machining technology has become an important development direction of modern manufacturing science. At present, ultra-precision machining technologies mainly include ultra-precision cutting technology [3], ultra-precision grinding technology [4], and ultra-precision polishing technology [5]. However, the ultra-precision polishing technology is based on keeping the physical properties of the processed materials to achieve the ultimate surface quality, shape accuracy and dimension accuracy. It is one of the most effective processing methods to obtain high-quality workpiece surfaces efficiently [6-8].

Due to the unique advantages of ultra-precision polishing technology in processing, it has attracted the attention of scientific research institutions worldwide. So far, many new ultra-precision polishing methods have been proposed, including small grinding head polishing [9], laser polishing [10], consolidated abrasive polishing [11], chemical-mechanical polishing [12], abrasive water jet polishing [13], ion beam polishing [14], and magnetorheological polishing [15]. Among them, magnetorheological polishing technology is an emerging ultra-precision polishing technology with high precision of processed surface shape, wide machining range and small depth of subsurface damage, which is a hot topic in current research [16–20]. This technology is mainly used in ultra-precision machining, with high processing accuracy. The research content of magnetorheological polishing technology (various factors affecting the accuracy of magnetorheological polishing) includes the development process, polishing mechanism, processed materials, magnetic processing equipment, process methods, etc. Scholars at home and abroad have carried out systematic research on these contents. However, there are very few literature reviews on the research content of magnetorheological polishing technology. This also hinders the in-depth development of magnetorheological polishing technology. Therefore, an overview of the research content of magnetorheological polishing technology is a key basis for the further development of this technology in the future.

In this paper, the research progress of magnetorheological polishing technology is reviewed. Section 1 introduces the proposal and development of magnetorheological polishing technology. Section 2 summarizes the materials processed by magnetorheological polishing technology, focusing on their application and the processing accuracy that can be achieved. Section 3 introduces the research progress of magnetorheological fluid (MRF), mainly on the sedimentation stability and shear yield stress of MRF and the research progress of magnetorheological polishing fluid (MPF). Section 4 divides magnetorheological polishing techniques into wheel-type, clustertype, ball-type, disc-type, etc. It mainly introduces its polishing principle, equipment research and development, theoretical analysis and processing technology. Section 5 compares the above types of magnetorheological polishing techniques from the perspective of material removal rate, surface roughness, surface uniformity, and range of application. In addition, the curvature-adaptive magnetorheological polishing technology with a circulation system is introduced. Section 6 presents the future development trend of magnetorheological polishing technology prospects.

### 2 Proposal and development of magnetorheological polishing technology

The development history of magnetorheological polishing technology is shown in Fig. 1, which goes through the following three stages.

# 2.1 Period from the 1940s to 1970s being the beginning of technology

As early as 1948, researcher Rabinow [21] first discovered the phenomenon of the rheology of magnetic fluids in the presence of magnetic fields. The phenomenon is that a magnetic fluid behaves as a liquid with good fluidity under a zero magnetic field, and a magnetic fluid behaves as a solid-like characteristic under a strong magnetic field. After research, it was found that the phenomenon changed continuously and controllably, and the magnetic fluid was subsequently referred to as magnetorheological fluid. However, its development was prolonged in the following three decades because the rheological properties of magnetorheological fluids have yet to be recognized.

# 2.2 Period from the 1970s to the early twenty-first century being the development of technology

In 1986, Professor Kordonski pioneered the research on magnetorheological polishing technology, and confirmed that a rationally formulated magnetorheological fluid in a magnetic field could polish optical materials. Since then, magnetorheological polishing techniques have emerged [22, 23]. Since 1990, due to the booming development of medical devices, optoelectronics, and communications industries, people have had higher product quality, reliability, and performance pursuit. This forced the further development of magnetorheological polishing technology. In 1992, Prof. Kordonski designed the world's first prototype of magnetorheological polishing technology [24]. However, the prototype does not have the basic idea of computer-controlled optical surfacing (CCOS), which can only process aspherical surfaces now. It is difficult to trim the surface shape with high precision [24]. Subsequently, Professor Kordonski collaborated with the Center for Optics Manufacturing (COM) at the University of Rochester to develop magnetorheological polishing equipment [25]. In 1993, this team developed a new generation of magnetorheological polishing prototype. Through many experimental studies, it verified the high efficiency and low damage polishing advantage of magnetorheological polishing technology for optical materials [26]. In 1995, for developing cutting-edge technologies such as aerospace and defense, the COM was the first to develop orthorhombic magnetorheological polishing equipment. Moreover, a batch of optical components with a diameter of less than 50 mm was processed by it. A high-quality surface with a surface roughness  $R_{\rm a}$  of 1 nm was obtained, thus starting the innovation of magnetorheological polishing technology [27]. In 1996, the COM added the idea of CCOS to the original orthotropic wheel magnetorheological polishing equipment, which brought it into the era of intelligent computer numerical control (CNC)



Fig. 1 Development history of magnetorheological polishing technology

[28]. In 1997, the COM started commercializing magnetorheological polishing equipment by establishing QED. However, at this time, the application of magnetorheological polishing equipment has certain limitations, mainly in the form of special machines to order [29]. Since the beginning of the twenty-first century, research on magnetorheological polishing technology has blossomed everywhere. Many kinds of magnetorheological polishing techniques exist, such as wheel-type, cluster-type, ball-type, disc-type, etc. The typical teamson wheel-type magnetorheological polishing technology include the QED in the United States [30], Zhang's team at Changchun Institute of Optics, Fine Mechanics and Physics (CIOMP) [31–33], Feng's team at Tsinghua University [34], Peng's team at National University of Defense Technology [35–37], Schinhaerl's team at the University of Applied Sciences Deggendorf [38, 39], Cho's team at Inha University [40], Zhang's team at China Academy of Engineering Physics [41], etc. It is worth mentioning that the QED has quickly become the world's leader in magnetorheological polishing and related technologies. The typical teams on cluster-type magnetorheological polishing technology include Yan's team at Guangdong University of Technology, China [42]. The typical teams on ball-type magnetorheological polishing technology include Anant's team at the Indian Institute of Technology [42–45] and Chen's team at the Harbin Institute of Technology [46]. The typical teams on disc-type magnetorheological polishing technology include Yin's team at Hunan University [47].

# **2.3** Period from the early twenty-first century to the present being the application of technology

Since the early twenty-first century, a new global scientific and technological revolution and industrial change have developed. New technologies have been accelerating the integration with ultra-precision machining technology. As a new research direction of ultra-precision machining technology, magnetorheological polishing technology plays an important role in the fields of national defense aviation, optoelectronic communications, medical instruments and civil industry due to its unique polishing advantages. The followings are the main applications of magnetorheological polishing technology.

# 2.3.1 Defense and aviation, optoelectronic communication field

The development of modern warfare, defense and aviation technology depends on advanced optoelectronic communication, control and weapon equipment technology. However, advanced optoelectronic communication, control and weapon processing are inseparable from applying magnetorheological polishing technology. For example, using magnetorheological polishing for the convex spherical (BK7) mirror used in the nuclear fusion device can converge the root-mean-square (RMS) of the mirror from 22 nm to 6 nm, improving the performance of the mirror [48]. Using magnetorheological polishing for the hemispherical resonant gyroscope used in the Jetlink inertial guidance system allows obtaining a surface roughness of  $R_a$  3.2 nm, which improves the accuracy of the hemispherical resonant gyroscope in use [49]. Using magnetorheological polishing for the surface damage of fused silica optics used in high-power lasers, a 90% surface damage repair rate can be obtained, effectively reducing the waste of fused silica material [50].

### 2.3.2 Medical instruments field

The application of biomedical materials has greatly contributed to the development of medical standards and has the enormous market potential [51]. However, for biomedical materials, ultra-smooth surface quality and ultra-low surface roughness are important constraints to their development. Therefore, applying magnetorheological polishing technology is indispensable to obtain a high-quality machined surface. For example, the use of magnetorheological polishing for ultra-high molecular weight polyethylene (UHMWPE) is used in the artificial acetabulum, which increases its microhardness and improves its service life accordingly [52]. Using magnetorheological polishing for the titanium alloy used in artificial joints, ultra-smooth surface quality with a surface roughness  $R_a$  of 2.87 nm can be obtained, improving its service performance [53]. Using magnetorheological polishing for the cobalt-chromium-molybdenum alloy used in the artificial hip joint allowed obtaining a high-quality surface of 5 nm, improving its accuracy [54].

### 2.3.3 Civil industry field

With the progress of science and technology, pursuing highquality, high-performance, and high-reliability products has greatly promoted the application of magnetorheological polishing technology in the civil industry. For example, magnetorheological polishing for stainless steel parts formed by selective laser melting (SLM) can effectively reduce the residual stresses on their surfaces [55]. Using magnetorheological polishing for the inner wall of a hydraulic cylinder used by heavy machinery can effectively improve its surface accuracy and increase its service life [56]. Using magnetorheological polishing for the cylindrical copper rod in the printing press can reduce its surface roughness and enhance its surface characteristics [57].

### 3 Magnetorheological polishing materials

Once the magnetorheological polishing technology was proposed, it attracted the attention of domestic and foreign researchers because of its unique advantages and ability to achieve low-loss processing. Numerous studies have shown that researchers are keen to conduct relevant research on new functional materials. At present, from the level of ultra-precision machining, the new functional materials in the existing research are divided into the following categories: soft and brittle artificial crystal materials, hard and brittle optical crystal materials, hard and brittle ceramic materials, and alloy materials [58, 59]. Table 1 lists the applications and precision of some novel functional materials in existing research [11, 53, 54, 60–69].

The physical properties of some new functional materials in existing research are listed in Table 2. Due to its superior physical and chemical properties, new functional materials are widely used in aerospace, medical devices, optoelectronic communications and the civil industry [62, 70]. However, these new functional materials have the same characteristics: high brittleness, high hardness and easy-to-produce sub-surface damage, which have an extremely adverse effect on the device's performance and working life. Meanwhile, it also challenges ultra-precision machining greatly [71, 72]. Therefore, to meet the application requirements of the novel functional materials, it is necessary to ensure their external dimensions, surface shape accuracy, surface roughness, and other precision requirements.

### 4 Research progress of magnetorheological fluid

#### 4.1 Research on properties of magnetorheological fluid

MRF is a liquid with rheological properties. When the liquid is in the magnetic field, it will have rheological behavior due to the guidance of magnetic particles. It is mainly composed of magnetic-sensitive particles, base carriers and additives. At present, the research on MRF in magnetorheological

Table 1	Applications a	and precision of	of some of the no	vel functional	l materials in	existing researches
---------	----------------	------------------	-------------------	----------------	----------------	---------------------

Classification	Example	Application	Precision/nm
Soft and brittle artificial crystal materials	CaF <sub>2</sub>	Extreme ultraviolet lithography	1.74 [11]
	ZnSe	Infrared optical instrument	1.57 [60]
	KDP	Electro-optical modulators, Q-switches	1.2 [61]
Hard and brittle optical crystal materials	К9	Lasers, LCD TVs	1.4 [62]
	SiO <sub>2</sub>	Astronomical telescopes, Lasers	0.165 [63]
	$Al_2O_3$	Infrared alert system, Missile fairing	0.27 [64]
Hard and brittle semiconductor materials	Si	Rectifiers, diodes	0.5 [65]
	InP	Integrated circuits	0.35 [66]
	SiC	Transistors, radar	0.49 [67]
Hard and brittle ceramic materials	$ZrO_2$	Ceramic armor	0.7 [68]
	Si <sub>3</sub> N <sub>4</sub>	Precision bearings, wear-resistant tools	4.35 [69]
Alloy materials	TC4	Heat shields, engine blades	2.87 [53]
	CoCrMo	Artificial bones	5.0 [54]

 Table 2
 Physical properties of some new functional materials in existing research

	$Al_2O_3$	SiC	${\rm Si_3N_4}$	SiO <sub>2</sub>
Density, $\rho/(\text{kg}\cdot\text{m}^{-3})$	3 950	3 200	3 500	2 200
Modulus of elasticity, E/GPa	350	410	310	77
Mohs hardness	9.2	9.5	9.0	6.5
Compressive strength, $\sigma_{\rm bc}$ /GPa	2-2.7	2-2.5	3.5	1.15
Bending strength, $\sigma_{\rm b}$ /GPa	0.3–0.5	0.45	0.6	0.089
Fracture toughness, $K_{\rm IC}$ / (MPa·m <sup>1/2</sup> )	3.00	3.50	5.30	2.12
Corrosion resistance	Strong	Strong	Strong	Ordinary

polishing technology mainly includes sedimentation stability, shear yield stress, etc.

#### 4.1.1 Sedimentation stability

The magnetic-sensitive particles, base carrier and additives are mixed in a certain proportion, and the mixed liquid formed after a long period of dispersion is MRF. However, the magnetic particles in the free state of the MRF will condense due to the interaction force between molecules. When more and more magnetic particles are condensed, sedimentation occurs. However, the amount of magnetic particle sedimentation determines the stability of the MRF. It also determines the effective use time of the MRF. This is ultimately reflected in the processing efficiency and quality of polishing.

Xiong et al. [73] prepared MRF using oleic acid, polyethylene glycol and sodium dodecyl benzene sulfonate (SDBS) as surfactants. A new evaluation method for the anti-sedimentation stability of MRF was explored by measuring the influence of dispersed particles on the wettability of MRF. It is concluded that the smaller the HLB value of the surfactant, the better the anti-sedimentation performance of MRF. The contact angles of the three surfactants are shown in Fig. 2.

Shu et al. [74] modified hydroxy iron powder with nonionic surfactant polyethylene glycol (PEG) and obtained a water-based MRF after modification. The lowest sedimentation ratio is only 8.56%, and the initial performance is restored after re-stirring. The volume removal rate of K9 glass was higher than 1.4 mm<sup>3</sup>/min by using the prepared MRF. The SEM photos of hydroxyl iron powder before and after modification are shown in Fig. 3.

Yang et al. [75] used dimeric acid as a surfactant to prepare MRF and believed that the MRF containing dimeric acid formed loose flocculation and enhanced sedimentation stability. Niu et al. [76] prepared MRF using oleic acid and polymethyl methacrylate as surfactants. It is concluded that oleic acid and polymethyl methacrylate can improve the settling stability of MRF. This is because both can provide a stronger steric hindrance effect, as illustrated in Fig. 4.

Song et al. [77] used nano-carbonyl iron particles to partially replace micron-carbonyl iron in dispersing. It is concluded that the sedimentation stability can be improved by adding the proper amount of nanoparticles. Piao et al. [78] synthesized magnetic  $Fe_3O_4/SiO_2$  nanoparticles by sol-gel method. When added to MRF, the rheological properties and dispersion stability of MRF were higher than those of pure MRF, and the sedimentation rate was reduced by 28.3%. Guo et al. [79] prepared MRF with three kinds of nano-SiO<sub>2</sub> with different specific surface areas (150 m<sup>2</sup>/g, 200 m<sup>2</sup>/g, 380 m<sup>2</sup>/g) and carbonyl iron powder. It is concluded that SiO<sub>2</sub> will gelate after entering MRF, which will increase the viscosity of MRF. To a certain extent, the fluidity of the MRF



Fig. 2 Contact angles of the three surfactants [73] a polyethylene glycol, b oleic acid, c SDBS



Fig. 3 SEM photos of hydroxyl iron powder before and after modification [74] a hydroxy iron powder, b PEG-200, c PEG-400, d PEG-600, e PEG-800, f PEG-1000, g oleic acid

is reduced, thus slowing down the settlement. The results obtained after the static experiment are shown in Fig. 5.

In summary, to improve the settling stability of MRF, it is necessary to add different kinds of additives to MRF. The additives added to MRF in existing studies are represented by oleic acid and SiO<sub>2</sub>, but the principles of action of the two are also very different [80]. Oleic acid, by coating the magnetically sensitive particles to form a wrapping structure, blocks the interaction between the iron powders, slowing the settlement. SiO<sub>2</sub>, the gelation of SiO<sub>2</sub> will form a stable three-dimensional structure, which plays a certain buffer role in the aggregation of magnetic particles, thereby slowing down the sedimentation rate.

#### 4.1.2 Shear yield stress

The shear yield stress of MRF refers to the minimum stress value required for MRF to change from a flow state to a solid state under the action of the magnetic field. Shear yield stress is an important property index of MRF and one of the important parameters for measuring the rheological properties of MRF.

Zhu et al. [81] prepared four kinds of MRFs with different volume fractions (10%, 20%, 30%, and 40%) to study their shear properties. The shear stress of these four MRFs at different shear rates was measured by the rheometer, as shown in Fig. 6. It is considered that the larger the volume



Fig. 4 Diagram of steric hindrance effect of oleic acid and polymethyl methacrylate [76]



Fig. 5 Results obtained after static experiment [79]

fraction, the better the shear performance of MRF. Xiao et al. [82] discussed the microscopic changes of silicone oil-based MRF under the action of a magnetic field (see Fig. 7) and the corresponding shear mechanical properties. It is concluded that the shear stress of MRF depends on the strength of the applied magnetic field, and there is a certain relationship between them. When the current is small, the shear stress increases exponentially, and the index value is about 1.5. When the impressed current increases, the shear stress increases linearly and finally reaches a stable value.

Ginder and Davis [83] explored the relationship between the diameter of magnetically sensitive particles and shear stress. It is concluded that the shear yield stress of MRF



Fig. 6 Effect of shear rate on shear stress of MRFs [81]

increases significantly with the increase in particle size. However, when the particle size exceeds a certain range, the shear yield stress of MRF is independent of the particle size. Zhao et al. [84] studied the microstructure and shear properties of the interface between carbonyl iron particles (CIPs) and dimethylsilicone oil (DSO) by molecular dynamics simulation method. It is concluded that strong bonding strength exists between the Fe atom on the CIPs surface and DSO molecular chain during the shearing process. MRFs with different viscosity DSOs were prepared. It was found that high interfacial shear strength could improve the rheological properties of MRFs. Zhu et al. [85] established a multi-chain calculation model of MRF based on the magnetic interaction energy between particles. The magnetic shear stress-strain relationship is obtained using the magnetic energy density variation. Li [86] obtained the relationship between the shear yield strength and magnetic field strength of IL-MRF and SO-MRF based on Bingham plastic model fitting, as shown in Fig. 8. It can be seen from Fig. 8 that the shear yield strength of IL-MRF is greater than that of SO-MRF under a larger external magnetic field. It is worth noting that when the applied magnetic field reaches 436 kA/m, the shear yield strength of IL-MRF is 29% higher than that of SO-MRF. That is, IL-MRF exhibits a better magnetorheological effect.

In summary, the microstructure of shear stress in MRF is as follows: under the action of a magnetic field, the magnetically sensitive particles gradually form a chain structure and are in a condensed state. When the magnetization of the magnetically sensitive particles continues to increase, the shear stress also changes, showing a trend of obvious increase. In addition, the particle size and magnetic field strength of magnetically sensitive particles greatly influence the shear stress.



Fig. 7 Microscopic changes of MRFs with different magnetic field intensities [82] a 0 mT, b 15 mT, c 75 mT, d 150 mT



**Fig. 8** Relationship between the shear yield strength of IL-MRF and SO-MRF and the magnetic field strength [86]

# 4.2 Research status of magnetorheological polishing fluid

The main components of MPF are similar to those of MRF. It is mainly composed of base carrier fluid, magnetic particles, abrasive particles and surfactants. Currently, MPF can be divided into two types according to the different base carrier fluids. The first is a water-based MPF using water as the base carrier fluid. It can be used to process the polishing of most materials. However, it is not able to process watersoluble materials. The second is the oil-based MPF using oil (mineral oil, silicone oil, etc.) as the base carrier fluid. It is suitable for the processing of water-soluble materials. As an important part of magnetorheological polishing technology, the properties of MPF directly affect the processing quality of optical materials. Scholars at home and abroad have conducted extensive research on it.

Bai [87] used zirconia to modify the surface of carbonyl iron powder to study the properties of MPF prepared by this composite particle. It is concluded that the modified carbonyl iron powder with MPF has better stability and a higher material removal rate for polishing glass and ceramic crystals. Mahender et al. [63] investigated the influence of temperature on the properties (shear force and viscosity) of water-based MPF. It is believed that the higher the temperature, the smaller the shear force and viscosity of the MPF. Subsequently, a water-based MPF with cerium oxide as abrasive was prepared at 25 °C, and a high-quality surface of 0.165 nm was obtained by polishing the fused quartz glass. Bai et al. [88] from Changchun Institute of Optical Machinery configured a water-based MPF suitable for optical processing. K9 and Si were polished with the prepared polishing fluid, and the material removal rates of 4.83 µm/ min and 1.376 µm/min were obtained, respectively. The stability and high removal efficiency of the prepared polishing fluid were verified. Peng et al. [61] from the National University of Defense Technology developed an oil-based MPF. The removal function experiment of KDP crystal is carried out, and the stable removal function is obtained through optimization.

With the in-depth study of the performance of MPFs, MPFs are gradually commercialized. OED Company in the United States has developed a variety of commercial MPFs, such as QED-C11, QED-D10, QED-D20, QED-C30, etc. These MPFs have low zero-field viscosity, high material removal efficiency, and stable composition under the control of a recirculation system. They are now used in practical engineering processes. According to available data [89], the PRR (peak removal efficiency) for polishing BK7 by QED-C10 can reach 4.47 µm/min, and the PRR for polishing single crystal Si by QED-D20 can reach 5 µm/min. However, QED is absolutely secretive about the polishing fluid's composition, content, and formulation process, and the product is expensive. There are also many institutions in China to develop a series of MPFs. For example, ZJY-C-A/B water-based MPF developed by Hunan Norbest Technology Co., LTD. It is mainly used for the magnetorheological polishing of quartz, microcrystal, K9, metal mirror and other materials. ZJY-D-A/B water-based MPF is mainly used for the magnetorheological polishing of crystals, hard and brittle materials, ceramics and sapphire. However, the domestic research on the preparation process and additive composition of MPF is not deep enough. The stability of fluid and material removal efficiency still have increased gaps with foreign countries.

# 5 Research status of magnetorheological polishing technology

### 5.1 Wheel-type magnetorheological polishing technology

#### 5.1.1 Principle introduction

The wheel-type magnetorheological polishing technology uses the "point contact" processing method to process the workpiece. It is mainly used in the high-efficiency and highquality processing of aspherical surfaces, spherical surfaces, complex surfaces, large-size prism, and other components. It can obtain sub-nano-level polishing quality. Depending on the different structures of polishing equipment can be divided into upright type and inverted type. Take the inverted magnetorheological polishing equipment as an example, its principle is shown in Fig. 9a [90]. The magnetic poles are located inside the polishing wheel. When the electromagnet is energized, a high-gradient magnetic field is generated between the magnetic poles and the workpiece to be processed. A cycle system transports the magnetorheological fluid to the surface of the polishing wheel. When the magnetorheological fluid passes through the high-gradient magnetic field, it forms a flexible ribbon on the surface of the polishing wheel. When the flexible ribbon comes in contact with the workpiece and creates mutual motion, a tremendous shear force is generated between the two. Thus, the surface of the workpiece is micro-removed. The material removal function is shown in Fig. 9b [91].

#### 5.1.2 Equipment research and development

QED is a leader in the development of wheel-type magnetorheological polishing equipment. It has already produced many more mature magnetorheological polishing equipment and realized the commercialization of wheel-type magnetorheological polishing equipment. Among them, the



Fig. 9 Principle of wheel-type magnetorheological polishing and material removal function **a** working principle [90], **b** material removal function [91]

smallest polishing wheel has been developed with a diameter of 10 mm, which can effectively process small complex parts. The largest wheel developed has a diameter of 370 mm, which can process concave lenses with the maximum size of 2.3 m and convex lenses of 1.7 m. Some of the latest magnetorheological polishing equipment produced by the company is shown in Fig. 10. In order to increase the applicability of the equipment, Peng's team [92] from the National University of Defense Technology designed aninverted magnetorheological polishing machine based on the traditional wheel polishing equipment. Its four-axis linkage and spindle control can realize the correction of the surface profile of many optical parts. The equipment was used to process optical glass, and its Peak-to-Valley (PV) value was successfully reduced from 261.7 nm to 55.3 nm, RMS value from 46.8 nm to 5.5 nm, and surface roughness  $R_{\rm a}$ value from 1.58 nm to 0.57 nm. The equipment is shown in Fig. 11a. Further, Li et al. [90] at Changchun Optical Machinery Institute increased the flexibility of the polishing equipment by integrating the polishing equipment into a five-axis CNC machining center. A hardware basis is provided for the processing of large-aperture aspherical optical components. The equipment is used to process a large diameter SiC with a size of 2 m, and its RMS value successfully converges from  $0.098\lambda$  to  $0.019\lambda$ , which verifies the practicability of the equipment. The equipment is shown in Fig. 11b. Wang et al. [93] of Tsinghua University proposed a new wheel-type magnetorheological polishing equipment with the form of revolution and rotation motion, which realized the renewal of magnetorheological fluid by alternating the strength and weakness of the magnetic field area. The reliability of the equipment was also verified by polishing experiments on optically aspherical surfaces. The equipment is shown in Fig. 11c. In order to further reduce the production cost, Shu [94] of Southwest University of Science and Technology integrated the wheel-type magnetorheological polishing equipment into the robotic arm. It can not only increase the size of the workpiece but also reduce the cost and, at the same time, has a high processing accuracy and efficiency. The equipment is shown in Fig. 11d. Huang et al. [41] of China Academy of Engineering Physics, configured two magnetorheological polishing heads with a diameter of 50 mm and 430 mm, respectively, on the same 8-axis CNC machine tool. A series of process tests were carried out with the equipment; the material removal characteristics and surface shape correction ability of large and small grinding heads were investigated; and a good polishing effect was obtained. Its equipment is shown in Fig. 11e. In addition, the commercialization of wheel-type magnetorheological polishing equipment has also made certain progress in China.



Q-flex-300

Fig. 10 Some of the latest magnetorheological polishing equipment

Q-flex-100



Fig. 11 Wheel-type magnetorheological polishing equipment developed by some domestic research institutes **a** National University of Defense Technology [92], **b** Changchun Optical Machinery Institute [90], **c** Tsinghua University [93], **d** Southwest University of Science and Technology [94], **e** China Academy of Engineering Physics [41]

So far, many companies have produced commercial magnetorheological polishing equipments. For example, the DQ series produced by Changguang Daqi Technology Co., Ltd.; MRP series produced by Suzhou Langxin Precision Optics Co., Ltd.; ZJY-NCM series produced by Hunan Norbest Technology Co., Ltd. Table 3 describes the parameters of some commercial magnetorheological polishing equipment. Among them are 12 kinds of polishing heads for wheel-type magnetorheological polishing equipment, with the smallest size of 10 mm and the largest size of 370 mm. At present, the wheel-type magnetorheological polishing equipment primarily uses electromagnets as the magnetic field source. The above commercial wheel-type magnetorheological polishing equipment provides a solid hardware foundation for developing magnetorheological polishing technology.

#### 5.1.3 Theoretical research

5.1.3.1 Material removal model research In order to better understand the material removal mechanism of wheeltype magnetorheological polishing, the theoretical model of the magnetorheological polishing process was established. Miao et al. [95, 96] proposed the Preston equation for material removal by combining the shear force on the workpiece surface with the material's mechanical properties. They verified through experiments that the material volume removal rate in the polishing process had a linear relationship with the shear force. Sidpara and Jain [97] established a theoretical model of the shear force of magnetorheological fluid to study the influence of process parameters (rotation speed, working clearance, etc.) on the tangential force of the workpiece during polishing. The accuracy of the theoretical

 Table 3
 Some parameters of commercial magnetorheological polishing equipment

Partial company	Туре	Polishing head specification	Applicable material	Applicable surface	Processing precision
QED	Q-FLEX-100 Q-FLEX-300 Q22-600 Q22-1200 Q22-750P2	10/20/50/150 mm 50/370 mm	Optical glass, silicon carbide, sapphire, etc.	Lane, sphere, aspherical surface, part shape, etc.	<ul> <li>① RMS &lt; 1 nm</li> <li>② RMS &lt; 0.5 nm</li> </ul>
Suzhou Langxin Preci- sion Optics Co., Ltd	MRP-300 MRP-600 MRP-1000	70/200 mm 70/200/340 mm 200/340 mm	Microcrystal, ULE, BK7, monocrystalline silicon, SiC, etc.	Plane, sphere, higher order aspherical sur- face, etc.	① RMS < 10 nm ② RMS < 0.6 nm
Changguang Daqi Tech- nology Co., Ltd	DQ-8 DQ-5 DQ-Flip-300 DQ-RM-150	50/80/160/360 mm 50/80/160 mm	K9, fused quartz, ULE, silicon carbide, etc.	Plane, sphere, quadric surface, etc.	RMS≤5 nm
Hunan Norbest Technol- ogy Co., Ltd	ZJY-NCM-600 ZJY-NCM-800 ZJY-NCM-1000	50/100/200 mm 100/200/340 mm 200/340 mm	Quartz, microcrystal, ULE, sapphire, etc.	Plane, sphere, higher order aspherical sur- face, etc.	① RMS < 10 nm ② RMS < 0.6 nm

① is the case of batch processing; ② is the situation of laboratory processing.

model was verified by polishing experiments on monocrystalline silicon. Liu et al. [98] proposed a material removal rate model involving pressure and shear stress, verifying the model's effectiveness through experiments. Bai et al. [88] investigated the effect of variation of each removal function shape and removal amount on the material removal stability by making removal function spots. It was concluded that the volumetric material removal efficiency variation was better than 1% when processing K9 glass and Si. Further, Liu et al. [99] proposed an interactive regionalized modeling method for the impact function of aspheric optics, which was considered effective in obtaining the impact function of aspheric tools. The computational flow of their regionalized modeling method is shown in Fig. 12. Bai et al. [100] established the normal pressure and effective friction equations of the polishing region based on the continuum and dense particle flow theories. On this basis, a new magnetorheological material removal model was established. The new model reveals the generation mechanism of the shear force of the polished workpiece, and realizes the effective decoupling of the main process parameters that affect the material removal of MPF. In addition, the material removal mechanism of wheel-type magnetorheological polishing has also been investigated by the team of Schinhaerl et al. [101] at the University of Daigendorf, Germany, and the team of Guo et al. [102] Akita Prefectural University, Japan.

5.1.3.2 Trajectory planning and dwell time algorithm research Chen et al. [103] proposed a new method for measuring normal profile errors when polishing complex surfaces on machining trajectories. The workpiece's clamping accuracy, the multi-axis machining numerical control program, and the magnetorheological polishing equipment's dynamic performances are obtained by dynamic measurement of normal contour error. It provides a basis for the validation and safety check of the magnetorheological polishing process.



Fig. 12 Calculation flow of regionalized modeling method [99]

Schinhaerl et al. [104] proposed a new dwell time algorithm. The machining time is shortened and the polishing quality of the complex surface is improved. The results of the magnetorheological polishing series show that there is only a 5% deviation between the actual material removal characteristics of the polishing tool and the material removal characteristics expressed by the influence function. Xu et al. [105] proposed a local magnetorheological polishing removal function modeling method for complex surfaces, and solved the dwell time algorithm. The surface error PV value of complex curved surface is converged from  $0.216\lambda$  to  $0.033\lambda$  by experiments. It provides a new theoretical basis for wheel-type magnetorheological polishing mechanisms. Peng et al. [106] proposed a compensation machining method for complex surface magnetorheological polishing. The removal function of the spherical curvature closest to the spherical region on the complex surface was selected as the resident point removal function. The dwell time algorithm based on the linear equations model was solved. The dwell time distribution of each resident point and the surface shape residual were obtained by simulation, which improved the certainty and convergence rate of the magnetorheological polishing process. Hu et al. [107] converted the removal function matrix into the coefficient matrix of the linear equations solved by dwell time. The speed variation along the polishing path is obtained by fast iterative calculation based on the sparse matrix. Zhang et al. [108] converted the dwell time deconvolution operation into matrix operation, and used the least squares approximation and the best consistent approximation to solve the optimization model numerically. Li et al. [109] proposed a method for achieving deterministic machining of curve surfaces by using magnetorheological polishing technology under grating machining trajectories. This method is mainly achieved by combining a four-axis linkage machine tool with a variable removal function. The practicality of this method was verified through shaping experiments on a convex sphere with an aperture of 80 mm. Qian et al. [110] established a bow height error model for adjacent polishing points based on a linear grating polishing trajectory. Moreover, the variation pattern of bow height error was analyzed. Based on the variation law of bow height error, precise control of bow height error can be achieved, and processing quality can be improved. Huang et al. [111] proposed a polishing trajectory interpolation method based on an equal proportional feed rate adjustment strategy. This method can significantly improve the solving precision of dwell time and the convergence speed of shape errors in the polishing process. Gao et al. [112] proposed a dwell time optimization method based on particle swarm optimization. The method obtains the optimal selection of each dwell time point by judging the overall dwell time. Li et al. [113] proposed a fast dwell time algorithm based on the matrix computational model, combining the minimum nonnegative multiplication with adaptive regularization. This algorithm effectively solves the problem of high-precision magnetorheological machining of large-aperture optical components. In addition, Li et al. [91] also proposed a generalized surface error mapping extension algorithm applicable to any optical shape. This algorithm simplifies the MPF process effectively. Cheng et al. [114] established a step-by-step multilevel iterative trajectory error compensation method based on spatial similarity. This method can effectively compensate the trajectory error of the robot magnetorheological polishing, effectively improving the machining accuracy. Song et al. [115] proposed a universal post-processing algorithm for processing optical mirrors by grating scanning. Song et al. [116] proposed a constrained nonlinear model to solve the dwell time based on the interior point algorithm. It is considered that the finite and non-negative dwell time can be obtained in the range imposed by the CNC system and machine tool dynamics by using this model and algorithm. In order to suppress the edge effect in the polishing process, Dong et al. [117] proposed a modified dwell time optimization model through iterative and numerical methods. It is concluded that a smooth, continuous and non-negative dwell time graph and root-mean-square convergence rate of 99.6% can be obtained using this workpiece simulation model.

#### 5.1.4 Process research

With the importance of magnetorheological polishing technology in the technological revolution, the practicality of wheel-type magnetorheological polishing technology has been studied. Byung et al. [118] analyzed the effect of each factor on the polishing removal rate by designing a fivefactor, two-level orthogonal experiment. It was concluded that the magnetic field strength, carbonyl iron powder volume fraction, and polishing depth significantly affected the removal rate. Peng et al. [61] polished 80 mm × 80 mm KDP crystals by using wheel-type magnetorheological polishing equipment, and obtained a high-quality surface with a surface roughness of 1.2 nm. Li [25] used self-developed MRF360 wheel-type polishing equipment to polish 1.5 m scale off-axis aspherical surfaces and obtained a surface quality of 12.9 nm. The actual picture of its processing is shown in Fig. 13. Their polishing results are shown in Fig. 14. Liu et al. [119] studied the effects of wheel magnetorheological polishing technology on the removal effect of the sub-surface damage layer of quartz glass. After processing, they believed that sub-surface damage layer was reduced from 18 nm to 4 nm. The magnetorheological polishing technology proved to be able to remove the sub-surface damaged layer. Shi et al. [120] explored the inhibitory effect of wheel-type magnetorheological polishing technology on the "comet tail" defect on the surface of monocrystal silicon through experimental and simulation methods. They



Fig. 13 MRF360 wheel type magnetorheological polishing machine processing physical [25]

concluded that reasonable technological parameters could effectively suppress the "comet tail" defect.

In summary, wheel-type magnetorheological polishing technology is the most mature and widely used. It is mainly reflected in the following aspects: many polishing equipments developed which can be used for processing largesize, aspheric, and spherical components. It has a wide range of applications. The theoretical basis is perfect. The theoretical model of material volume removal rate and shear force, and the dynamic model of complex surface local area removal function are established, which has certain guiding significance for practical production. The machining stability is high. The ultra-precision polishing of optical materials such as monocrystal silicon, K9 glass, quartz glass and KDP crystal can be completed. The ultra-smooth, less/no damage machining surface can be obtained.

Although wheel-type magnetorheological polishing is considered as the most mature technology, it has some limitations. It is mainly reflected in the following aspects: the polishing wheel is in point contact with the workpiece when working. The number of abrasive particles involved in material removal per unit of time is small, which makes the material removal rate of the workpiece low. The polishing wheel has a single-motion track. There will be an overlap error between the two motion tracks, so the workpiece surface will have a certain error copy. Wheel-type polishing equipment requires high precision and is mainly made of large machines, which makes the equipment cost expensive. Due to the wheel's size limitation, only aspheric machining of apertures larger than 8 mm is currently possible [121]. Therefore, wheel-type magnetorheological polishing technology still has a certain development space. Further research can be carried out on material removal rate, surface quality, equipment cost and limit size machining.



Fig. 14 Face shape of large-aperture aspheric surface before and after polishing [25] a before polishing, b after polishing



Fig. 15 Principle of cluster-type magnetorheological polishing [122]

### 5.2 Cluster-type magnetorheological polishing technology

#### 5.2.1 Principle introduction

The processing feature of cluster magnetorheological polishing technology is "surface contact". Its main application is the processing of planar optical parts. It has high processing efficiency. Its processing principle is shown in Fig. 15 [122]. It embeds several permanent magnets uniformly under the polishing disc according to the cluster principle, thus forming a high-intensity gradient magnetic field. Several microabrasive heads are formed in the MPF under the action of the magnetic field. When the workpiece rotates with the spindle, shearing action occurs between the workpiece and the micro-abrasive heads, thus realizing the purpose of polishing the workpiece with high efficiency and flatness [42].

#### 5.2.2 Equipment research and development

Yan et al. [42] developed the first generation of clustertype magnetorheological polishing equipment. Its structure and physical diagram are shown in Fig. 16. However, the magnetic field of the first-generation polishing equipment is static, which will lead to the poor self-recovery ability of the magnetorheological polishing pad. Thus, the processing efficiency and uniformity of the workpiece are limited. In order to further improve the limitations of static magnetic field cluster magnetorheological polishing technology. Pan et al. [123] and Guo et al. [124] proposed that a dynamic magnetic field was formed by the synchronous eccentric rotation of magnetic poles in a cluster arrangement. Its physical picture is shown in Fig. 17. It can make the distorted polishing pad self-repair in realtime. It overcomes the disadvantages that the deformation of the polishing pad is difficult to recover, and the abrasive is accumulated under a static magnetic field, making the material removal process more stable and the polishing effect better. Further, Yan et al. [125] proposed a variable gap dynamic pressure cluster-type magnetorheological polishing method based on dynamic magnetic field cluster-type magnetorheological polishing technology. Using sapphire as the research object, an experimental study was conducted to compare the two. The experimental results show that this method can increase the material removal rate of the workpiece by 19.5% and reduce the



Clamping workpiece device 2. Rotary polishing disk 3. Machine body
 Recycling and storing device 5. Adjustable magnetic device





Fig. 17 Dynamic magnetic field cluster-type magnetorheological polishing equipment [123]

surface roughness by 42.96%. Experimental results show that the method is effective. Further, Luo et al. [126, 127] improved the polishing shear force by adding a round hole array to the polishing disc. Orthogonal experiments were used to optimize the size parameters of the round holes, which provided a new hardware basis for the cluster-type magnetorheological polishing technology. The structure and physical diagram are shown in Fig. 18. In addition, Lu et al. [128] of Changchun University of Technology also researched cluster-type magnetorheological polishing technology and developed reciprocating cluster-type magnetorheological polishing equipment. The physical picture of which is shown in Fig. 19.



Fig. 18 Circular hole array type dynamic magnetic field cluster-type magnetorheological polishing equipment [126]



Fig. 19 Reciprocating cluster-type magnetorheological polishing equipment [128] a device structure, b magnetic field generation device, c magnetorheological polishing film

#### 5.2.3 Theoretical research

Based on a series of magnetorheological polishing equipments developed by cluster-type magnetorheological polishing technology, many scholars have carried out a lot of theoretical research on material removal model, polishing force modeling, machining trajectory and so on. The Preston equation by introducing a correlation coefficient was modified in Refs. [129, 130]. Moreover, the workpiece polishing pressure was calculated based on the magnetic field distribution in the cluster-type magnetorheological polishing device. The material removal rate model for cluster-type magnetorheological polishing of silicon carbide was established. It provides theoretical guidance for cluster-type magnetorheological polishing technology. Pan et al. [131] studied the characteristics of the polishing force of the magnetorheological effect of a single-point dynamic magnetic field. It was concluded that relative to the static magnetic field, the polishing force and torque under the action of the dynamic magnetic field showed a dynamic behavior with large fluctuations, which enabled the renewal of abrasive particles in the MPF. Pan et al. [123] studied the influences of various process parameters on the polishing pressure of the workpiece surface based on fluid mechanics and the Preston equation. The material removal rate model of cluster-type magnetorheological polishing under a dynamic magnetic field was established to provide theoretical guidance for cluster-type magnetorheological polishing technology. Yan et al. [132] analyzed the mechanism of action of variable gap dynamic pressure cluster-type magnetorheological polishing. It argued that a structurally solid string of bodycentered tetragonal magnetic chains was formed under the axial squeezing vibration of the workpiece, which generated



**Fig. 20** Structural change of MPF under the action of variable gap **a** no applied magnetic field, **b** chains formed after applying a magnetic field, **c** the deformation of the magnetic chain string after extrusion, **d** the chain is broken and reassembled [132]

greater polishing pressure and abrasive grain binding force. It promotes the reflow renewal of abrasive particles. The material removal efficiency and machining surface quality were improved. The structural changes of MPF under the action of variable clearance are shown in Fig. 20. Luo et al. [133] established the shear force and material removal rate models of solid particles in MPF based on tribological principle. It provides theoretical guidance for the magnetorheological polishing of dynamic magnetic field clusters with circular hole arrays. Luo et al. [134] established a fluid



Fig. 21 Numerical model of magnetic track strength [135]

simulation model based on clearance variation MPF to evaluate the influences of changing machining clearance on MPF fluid flow field behavior. It was concluded that MPF with different gaps could improve the polishing efficiency and surface quality by changing the behavior of the flow field. Yan et al. [135] proposed a numerical model of magnetic track strength that included both effective machining track and magnetic track characteristics. This model can effectively optimize the polishing process parameters and improve the machining non-uniformity of the workpiece after polishing. The model establishment diagram is shown in Fig. 21. Zhuang et al. [136] established a material removal model related to the normal pressure, particle size, particle concentration, and the motion path length and verified the model's accuracy through experiments. Subsequently, Zhang et al. [137] optimized the material removal model of fused quartz glass by convolving the dwell time after the Fourier transform with the material removal function. The accuracy of the material removal model is further improved.

#### 5.2.4 Process research

Based on the polishing equipment and related theoretical basis developed by cluster-type magnetorheological polishing technology, many scholars have explored its process. Yan et al. [138] studied the influences of cluster magnetorheological process parameters (polishing pressure, processing speed, etc.) on the processing effect of K9 glass. The experimental results showed that the polishing method could effectively improve the processing efficiency of the workpiece, which verified the practicability of the polishing method. Bai et al. [139, 140] found abrasive particles' "tolerance" effect



Fig. 22 Diagram of the "tolerance" effect [139]

in cluster-type magnetorheological polishing. The effect is shown as follows: only the top of the abrasive particles of different sizes in the MPF has contact with the surface of the workpiece, which improves the effect of ultra-smooth flattening of the machined surface accordingly. The schematic diagram of the "tolerance" effect is shown in Fig. 22. Pan et al. [141] conducted polishing experiments on SrTiO<sub>3</sub> substrate material, and obtained a high-quality machined surface with a surface roughness of 8 nm and high material removal efficiency of 0.154 µm/min. Pan et al. [123] polished the monocrystalline silicon substrate with a designed three-piece tool head for 4 h, and obtained a super-smooth surface with a surface roughness  $R_a$  of 2.464 5 nm. The surface morphology of single crystal silicon before and after polishing is shown in Fig. 23. Further, Pan et al. [64] optimized the process parameters (processing gap, workpiece speed, magnetic pole speed, etc.) for multi-pole dynamic magnetic field polishing of sapphire by orthogonal experiments. The optimal process parameters obtained an ultrasmooth surface with  $R_a$  0.27 nm. Yan et al. [142] and Liang et al. [143] polished single-crystal SiC by adding chemical reagents into MPF. It is believed that the material removal mechanism of SiC is as follows: chemical agents chemically corrode the SiC surface to form the corrosion layer. Abrasive particles remove the corrosion layer of single crystal SiC surface through mechanical action. The whole process is gradual. Lu et al. [20] optimized the magnetic field distribution of 2 mm above the electromagnet by reducing the angle of the iron core and opening annular slots with different edge angles to weaken the edge effect of the magnet. The central magnetic field intensity of the optimized electromagnet is greater than 300 mT, and the maximum magnetic field



Fig. 23 Surface morphology of single crystal silicon before and after polishing [123] **a** before polishing, **b** after polishing

intensity is 420 mT. It provides an effective optimization method for magnetic field distribution. Zhuang et al. [136] designed a curved permanent magnet with a curvature radius of 55 mm, which effectively attenuated the influence of linear velocity on polishing. After experimental verification, the curved permanent magnet can reduce the average surface roughness from 286 nm to 16 nm and the standard deviation of surface roughness to 0.92.

In summary, cluster-type magnetorheological polishing technology uses several permanent magnets to form several polishing micro-grinding heads to achieve the purpose of high-efficiency flattening and polishing of planar workpieces, which is the more studied magnetorheological polishing technology at present. It is mainly reflected in the following aspects. The development of the polishing device has gone through four stages: static magnetic field, dynamic magnetic field, variable gap dynamic pressure, adding a round hole array on the polishing disc. It already has a piece of more mature polishing equipment. The theoretical research is complete. The shear force model and material removal rate model of solid particles in the MPF are established. The characteristics of the polishing force of the magnetorheological effect of a single-point magnetic field are analyzed. The technology adopts the idea of "surface contact" to increase the contact area between the abrasive particle and the workpiece. So the material removal rate of the workpiece increases. The dynamic magnetic field method is adopted to enhance the self-recovery ability of the polishing pad, which is beneficial to the uniformity of the machined surface.

Although there is much research on cluster magnetorheological polishing technology, some limitations exist. It is mainly reflected in the following aspects: the shape and size of the workpiece are limited to a certain extent by using the processing mode of the workpiece disk and the throwing disc under. The MPF is placed in the throwing disc, which not only limits the circulation of MPF, but also affects the rotation speed of the throwing disc. To a certain extent, the processing efficiency and stability of the workpiece are affected. Therefore, cluster-type magnetorheological polishing technology still has a certain development space. Further research can be carried out to improve the device's applicability and increase the circulation system of MPF.

#### 5.3 Ball-type magnetorheological polishing technology

#### 5.3.1 Principle introduction

The processing characteristic of ball-type magnetorheological polishing technology is "point contact". It is mainly used to machinate concave aspheric surfaces and free-form surface optical parts with small curvature radius. High surface quality can be obtained. Its processing principle is shown in Fig. 24. The MPF enters the polishing area from the top of the polishing head through the closed magnetic duct. After being magnetized by the gradient magnetic field generated by the excitation device, a spherical solid-like flexible polishing film is formed on the surface of the polishing head. The polishing film is driven by the rotating polishing head to produce a shearing motion with the workpiece surface to remove the workpiece material.

#### 5.3.2 Equipment research and development

Singh et al. [44] from the Indian Institute of Technology developed solid polishing equipment based on the balltype magnetorheological polishing technology schematic diagram. The electromagnet is used as an exciting device for polishing equipment. The polishing force of MPF on the workpiece surface is controlled by adjusting the size of the coil excitation current, and the required machining surface quality is obtained. The physical polishing equipment is shown in Fig. 25. Subsequently, Sidpara and Jain [144] improved the ball-type magnetorheological polishing equipment. They replaced the electromagnet with a permanent magnet, making the polishing equipment more convenient and practical. Su et al. [46] from the Harbin Institute of Technology also researched on ball-type magnetorheological polishing technology. They developed a small-diameter permanent magnet ball head magnetorheological polishing



Fig. 24 Principle of ball-type magnetorheological polishing technology [43]



Fig. 25 Physical of the ball-type magnetorheological polishing equipment [44]

prototype, as shown in Fig. 26. Compared with the polishing equipment developed in Refs. [44], the prototype adopts a permanent magnet as the excitation device of polishing equipment. It adopts a four-axis linkage control mode, which makes the polishing head more flexible and can polish the workpiece of various curvations. In addition, the ball-type magnetorheological polishing for small aperture aspherical elements was also researched in Refs. [145–147]. The inclined axis magnetorheological polishing equipment was made, as shown in Fig. 27. The polishing principle is similar to that of the Harbin Institute of Technology, which can effectively avoid interference between the polishing head and the workpiece during processing. Further, Wang et al. [148, 149] of the Harbin Institute of Technology proposed an ultrasonic-magnetorheological composite polishing method. Its processing feature is to increase the contact efficiency of MPF and the workpiece surface through ultrasonic vibration, which effectively improves the workpiece processing efficiency and polishing quality. The developed polishing device is shown in Fig. 28 [150].

### 5.3.3 Theoretical research

Based on a series of magnetorheological polishing equipment developed by ball-type magnetorheological polishing technology, many scholars have carried out a lot of theoretical researches on material removal models, polishing force modeling, subsurface damage, etc. To better understand the relationship between abrasive grains and hydroxylated iron powder particles in a magnetic field. Zafar and Sunil [151] proposed a cell with a body-centered cubic (BCC) structure. They established a theoretical model for surface roughness during ball-type magnetorheological polishing, which provided a new theoretical basis for its mechanistic study. The BCC structure is shown in Fig. 29. Xu et al. [152] established a three-dimensional material removal model for the magnetorheological polishing zone of a small-diameter permanent magnet ball head by combining the analytical methods of lubrication fluid dynamics of sliding bearings. It is considered that the workpiece surface material removal is dominated by shear force removal. Chen et al. [153] performed a hydrodynamic analysis of the magnetorheological polishing zone of a smalldiameter permanent magnet ball head. The primary analytical relationships among shear stress and apparent viscosity, flow velocity and dynamic fluid pressure were obtained, as shown in





Needle tubing

Small permanet magnetic ball-end tool

Fig. 26 Small-diameter permanent magnet ball head magnetorheological polishing prototype [46]



Fig. 27 Inclined axis magnetorheological polishing equipment [145]



Fig. 28 Ultrasonic-magnetorheological composite polishing equipment [150]

Eq. (1). Wang et al. [154] established a material removal function model for ultrasonic-magnetorheological composite polishing and carried out experimental validation, which laid the theoretical foundation for its extension. Liu et al. [155] compared the effect of ultrasonic-assisted polishing with conventional polishing on the surface and subsurface damage of SiCp/Al composites by finite element simulation. It is believed that ultrasonic assistance can reduce the surface and subsurface damage of SiCp/Al composites.

$$\tau = \begin{cases} \eta_0 \gamma + +\operatorname{sign}(\gamma)\tau_0, & |\tau| > \tau_0, \\ 0, & |\tau| < \tau_0, \end{cases}$$
(1)

where  $\tau$  is the shear stress;  $\eta_0$  is the viscosity;  $\dot{\gamma}$  is the flow rate;  $\tau_0$  is the fluid dynamic pressure.

#### 5.3.4 Process research

Based on the polishing equipment developed by the balltype magnetorheological polishing technology and related theoretical basis, many scholars have explored its process. The related processes based on this equipment was studied in Refs. [156-160]. The polishing force of the MPF on the surface of the workpiece is controlled by adjusting the magnitude of the excitation current of the coil. The surface roughness  $R_{a}$  of 16.6 nm was obtained by polishing the planar workpiece of ferromagnetic material. Typical three-dimensional surfaces were polished to obtain a surface quality of  $R_{\rm a}$ of 19.7 nm. BK7 optical glass and copper alloy surfaces were polished with a surface roughness  $R_a$  of 22 nm and 28.8 nm, respectively. Further, Sidpara and Jain [161] realized the polishing of the surface of complex free-form knee joint by driving a three-axis CNC milling machine, and the optimal surface roughness  $R_a$  was 28 nm. Tian et al. [162] explored the effect of MPF temperature on the material removal rate by developing a hydrodynamic model of the polishing region. It was concluded that increasing the temperature could significantly improve the polishing efficiency and polishing quality. The processing performance is shown in Fig. 30. Singh et al.



Fig. 29 BCC structure [151]



Fig. 30 Effect of different MPF temperatures on the process performance of permanent magnet ball head magnetorheological polishing [162] **a** material removal depth at different temperatures, **b** surface quality of the workpiece before and after polishing at a temperature of 60  $^{\circ}$ C

[163, 164] used response surface methodology to investigate the effect of process parameters on the percentage reduction of residual stress and surface roughness on the workpiece surface after polishing, which provided a theoretical basis for reducing the residual stress and surface roughness on the workpiece surface. Mayank and Pulak [165] investigated the contribution ratio of different process parameters (polishing speed, abrasive concentration and ultrasonic power) to the removal rate of single crystal silicon material during ultrasonic-magnetorheological composite polishing by response surface experiments. It concluded that ultrasonic power had the highest contribution to MRR at 57.9%, followed by polishing speed (13.3%) and abrasive concentration (12.5%). Hu [166] compared the effects of magnetorheological-ultrasonic composite polishing and magnetorheological polishing on the processing performance of zirconia ceramics. It was concluded that magnetorheological-ultrasonic composite polishing could effectively enhance the material removal rate of zirconia ceramics, but the surface roughness did not change significantly. The processing performance comparison is shown in Fig. 31. In order to reduce the error brought by the traditional measurement method, Lqbal et al. [167] realized in-situ measurement of the workpiece during machining by integrating a focusing sensor into the ball-type magnetorheological polishing equipment. They verified the feasibility of this method through experiments.

In summary, the main target of ball-type magnetorheological polishing technology is a small aperture and irregular optical component. Scholars at home and abroad have carried out a lot of researches. It is mainly reflected in the following aspects: the structure of the polishing device is simple and the polishing head is flexible. It is not limited by the shape and size of the workpiece and has a wide range of applications. The theoretical research is sufficient. The corresponding theoretical models of surface roughness and material removal are established, which can be used to guide practical production. The technology process is relatively mature, and there are corresponding studies on ferromagnetic materials, copper alloys and glass.

Although this technology has unique advantages in processing special-shaped parts and small diameter workpieces, it still



Fig. 31 Comparison of the processing performance of magnetorheological-ultrasonic composite polishing and magnetorheological polishing [166] a material removal, b surface roughness

has certain limitations. It is mainly reflected in the following aspects: due to the point contact between the polishing head and the workpiece, the material removal rate of the workpiece is low. When the workpiece is processed with complex opposites, a more precise control system is needed to control the precise displacement of the polishing head, increasing the polishing process's difficulty. Therefore, ball-type magnetorheological polishing technology still has a certain development space. Later, the polishing head can be fixed on the mechanical arm to increase its flexibility and controllability.

#### 5.4 Disc-type magnetorheological polishing technology

#### 5.4.1 Principle introduction

The processing characteristic of disc-type magnetorheological polishing technology is "surface contact". It is mainly used in the machining of large aperture planar optical parts. It has high processing efficiency and quality. Its processing principle is shown in Fig. 32. The magnetic field gradient is controlled by adjusting the excitation device in the polishing process, thus creating a large rectangular hardened area on the surface of the polishing disc. With the rotation of the workpiece spindle and polishing disc, shearing motion occurs between the workpiece surface and the large polishing tool, thus achieving the removal of material from the workpiece surface [168].

#### 5.4.2 Equipment research and development

Based on the principle of Fig. 32, Yin et al. [169] built a magnetorheological plane polishing equipment with a large polishing mode, as shown in Fig. 33a. The equipment adopts the polishing method of workpiece on the top and polishing



Flow direction Z Y

Fig. 32 Principle of magnetorheological polishing of large polished tool [47]



Fig. 33 Large polishing mold magnetorheological polishing equipment a permanent magnet [169], b electromagnet [68]

disc under, and uses permanent magnet as the excitation device. The polishing experiment of K9 glass verifies the practicability of the technology. However, the permanent magnet's magnetic field strength and distribution area are limited, which makes the polishing efficiency and accuracy need to be further improved. Guo et al. [68] improved the excitation device of magnetorheological polishing equipment for large polishing mold, as shown in Fig. 33b. The direct current electromagnet is used instead of a permanent magnet. The structure of the yoke is improved, and the structure of the pole of the array with sharp teeth and curved edges is designed to improve the problem of the permanent magnet. However, whether it is a permanent magnet or an improved direct current electromagnet, it produces a static magnetic field. Only when the MPF passes through the magnetic field processing area can it harden and make the abrasive move. In contrast, the abrasive is easy to accumulate in other areas without magnetic field. Therefore, Wu et al. [170] proposed using low-frequency alternating current as an electromagnet to stimulate the generation of the dynamic magnetic field. The magnetic field line of the dynamic magnetic field is constantly changing, thus promoting the uniform distribution of abrasive particles. The magnetic field motion mode and polishing effect are shown in Fig. 34. In order to further increase the size range of the workpiece to be processed, top-disc magnetorheological polishing based on the methods of small grinding head polishing and large polishing mode plane polishing was proposed in Refs. [171, 172]. Its principle is shown in Fig. 35a, and its processing effect is shown in Fig. 35c. It is worth mentioning that the equipment is equipped with an MPF circulation system, which is the first time that the MPF circulation system is applied to the disc polishing technology. It further promotes the development of technology in the direction of practicality and polishing stability.

### 5.4.3 Theoretical research

Based on the magnetorheological polishing equipment developed by disc-type magnetorheological polishing technology, many scholars have carried out relevant theoretical research on material removal model, polishing force modeling, sub-surface damage, etc. Wang et al. [173] established the motion trajectory equation of the workpiece on the magnetorheological hardening polishing tool, and based on the trajectory equation, the mathematical model of material removal was established. It lays a theoretical foundation for applying large polishing mode plane magnetorheological polishing. Guan et al. [174] established a workpiece material removal model and surface roughness prediction model based on self-developed polishing equipment. They conducted theoretical research on key parameters' influence on material removal efficiency and stability. Further, Luo [175] investigated the mechanism of sub-surface damage generation in the magnetorheological polishing of large polished mode. A mathematical model for damage layer depth prediction was developed and experimentally verified. A highquality surface with sub-nanometer surface roughness and only a few tens of nanometer damage layers was obtained. The subsurface damage layer is shown in Fig. 36.

#### 5.4.4 Process research

Based on the polishing equipment and related theoretical basis developed by disc-type magnetorheological polishing technology, many scholars have explored its process. Wang et al. [176, 177] conducted polishing experiments on K9 glass using a self-designed permanent yoke large polishing tool magnetorheological polishing equipment. An efficient and high-quality polish with the maximum volume removal rate of 0.794 mm<sup>3</sup>/min, a face shape accuracy PV of 1 µm and a



Fig. 34 Magnetic field motion pattern and polishing effect [170] a magnetic field motion pattern, b polishing effect



Fig. 35 Disc-type magnetorheological polishing technology [172] a principle of disc-type magnetorheological polishing, b before polishing, c after polishing



Fig. 36 Subsurface damage layer [175]

surface roughness of 0.6 nm was obtained. The effects of DC electromagnet magnetorheological polishing process parameters (polishing time, working gap and workpiece speed) on the machining properties of zirconia ceramics and monocrystal silicon were studied in Refs. [15, 68]. It is considered that under the optimal process parameters, a super-smooth surface with  $R_a$  of 0.7 nm can be obtained for zirconia ceramics, and a super-smooth surface with  $R_a$  of 0.4 nm can be obtained for monocrystalline silicon. The processing effect diagram is shown in Fig. 37. Jian et al. [65] composed a complex MPF by adding an oxidant and catalyst. The single crystal Si was polished with the composite polishing solution, and a highquality surface of  $R_a$  0.5 nm was obtained. Yang et al. [178] optimized the process parameters (polishing pressure, polishing disc speed, etc.) of magnetorheological polishing of sapphire disc by orthogonal method. It was concluded that a surface roughness  $R_a$  of 0.31 nm and a material removal rate of 2.68 µm/h were obtained under the action of the optimal process parameters. Yin et al. [179] prepared composite polishing liquid with different reaction systems based on the principle of self-hydration reaction of sapphire and the characteristics of chemical reaction with silica sol. It concluded that the MPF mixed with silica sol and  $\alpha$ -Al<sub>2</sub>O<sub>3</sub> abrasive had the highest removal efficiency, and obtained an ultra-smooth surface with *R*<sub>a</sub> of 0.3 nm.

In summary, the technology is mainly aimed at the processing of large aperture planar optical parts, and has achieved certain processing results. Among them, Hunan University has carried out the most relevant research. The technology uses a large polishing mode to contact the surface of the workpiece, effectively improving processing efficiency. The technology has also done a lot of exploration in the process, and has accumulated certain experience in difficult materials such as monocrystalline silicon and zirconia. However, this technology also has some limitations, mainly reflected in the technology for the large diameter curved components processing is more



**Fig. 37** Processing effects of magnetorheological polishing with DC electromagnet **a** surface quality of monocrystalline silicon before and after polishing [15], **b** surface quality before and after zirconia polishing [68]

difficult. Therefore, there is still some room for development of this technology. Subsequently, the five-axis CNC system can be added to the disc-type magnetorheological polishing equipment, so that it can adapt to the machining of large-diameter curved components.

### 5.5 Other forms of magnetorheological polishing technology

# 5.5.1 Rotation-typemagnetorheological polishing technology

Currently, various finishing parts (including molds for steering rack housings and actuators) have high requirements for blind hole cavities with surface finishing during injection molding [180]. To reduce the frictional characteristics and improve the functional efficiency of the blind hole cavity, Talwinder and Anant [181] proposed a rotation-type magnetorheological polishing technology. It is mainly used in the machining of blind cavity parts. The processing principle is shown in Fig. 38. Cylindrical permanent magnets are fixed to the spindle and MPF is distributed around the permanent magnets to form the polishing head. The polishing head rotates and moves to remove the material from the inner and bottom surfaces of the blind cavity, which results in a high-quality machined surface.



Fig. 38 Rotation-type magnetorheological polishing [181] a working principle, b material removal mechanism



Fig. 39 Comparison of the machining effect between the inner wall and the bottom surface of the cylindrical blind hole [182] a inner wall, b bottom surface

Aggarwal and Singh [182] used this technology to polish the inner wall and bottom surface of a cylindrical blind hole, resulting in a 72% reduction in the inner wall surface roughness and 79% reduction in the bottom surface roughness, which verified the usefulness of the technology. The comparison of its processing effect is shown in Fig. 39. Further, Aggarwal and Singh [183] optimized the process parameters (working gap, current, workpiece speed, etc.) for the rotation-type magnetorheological polishing technology. The polishing of EN-31 CBH-type cavities was carried out under optimal process parameters and good polishing results were achieved. Sirwal and Singh [184] developed a theoretical model for predicting machined surface roughness. The correctness of the theoretical model was verified by polishing experiments on the side and bottom surfaces of blind-hole workpieces with two forms of polishing heads, while providing a theoretical basis for the development of the technology. Sunil and Auant [185] investigated the effect of the motion behavior of abrasive grains in magnetorheological fluid on the polishing effect. It was concluded that the abrasive grains' simultaneous rotational and reciprocating motion was the main factor in enhancing the polishing effect. Yahya et al. [186] proposed a vibration-assisted rotation-type magnetorheological polishing technology to further improve the processing efficiency of the rotation-type magnetorheological polishing technology. The effect of this technology



Fig. 40 Working principle and processing effect of vibration assisted rotation-type magnetorheological polishing technology [186]

on the processing efficiency of the aluminum alloy 2024 tube workpiece was experimentally investigated. The technology can effectively improve the processing efficiency of the workpiece and achieve a better polishing effect. The processing principle and processing effect are shown in Fig. 40.

In summary, rotation-type magnetorheological polishing technology removes material from the inner and bottom surface of the blind hole cavity parts by rotating and moving the polishing head up and down to obtain high-quality machined surfaces. This technology successfully overcomes the problems of low efficiency and poor quality of traditional processing technology. This technology has done much work in process research and theoretical analysis, providing a basis for the further development.

However, the technology has some limitations. It is mainly reflected in the following aspects: the polishing head of the technology is inserted into the cavity of the component, which makes it difficult to cycle the MPF. This technology has relatively high requirements for detection technology and precision control technology in the processing process, while the technical level of existing research is limited. Therefore, rotation-type magnetorheological polishing technology still has a certain development space. Further research can be carried out in the fields of MPF cycle update, detection technology and precision control technology.

#### 5.5.2 Belt-type magnetorheological polishing technology

The processing feature of the belt-type magnetorheological polishing technology is that the box magnetic pole is used as the exciting device to increase the curvature radius of the polishing area, which effectively improves processing efficiency [187]. It is mainly used in the machining of large-diameter flat parts. Its processing principle is shown in Fig. 41a. When working, the MPF is ejected from the nozzle by the peristaltic pump and driven to the working magnetic field area by the high-speed rotating belt. Under the action of a high gradient magnetic field, the MPF quickly forms a ribbon bulge. The relative motion of the ribbon bulge and the workpiece surface achieves material removal. The material removal function is shown in Fig. 41b.

Ren et al. [188] conducted an experimental process study on BK7 glass and SiC based on self-developed belt-type magnetorheological polishing equipment. The experimental results showed that the material removal rate reached 2.52 µm/min for BK7 glass and 1.22 µm/min for SiC. Compared with the existing magnetorheological polishing device, the device can effectively improve the removal efficiency by more than five times. Wang [189] improved the belttype magnetorheological polishing equipment based on the design of Ren et al. [188]. The problem that the MPF was prone to oxidation by heat was solved. A model of the ribbon magnetorheological removal function's distribution was obtained using the virtual ribbon bump theory and the Preston material removal rate equation. After practical experimental verification, it is concluded that the model can guide the design and selection of the ribbon magnetorheological removal function. Finally, the polishing equipment was integrated into a 2.5 m gantry machine, and a 235 mm SiC plane mirror was processed. The convergence efficiency reached 78%, which verified the belt-type magnetorheological polishing ability and high processing efficiency. The processing effect is shown in Fig. 42.

In summary, this technology increases the radius of curvature of the polishing area, so that the contact area between the workpiece and the MPF increases. That is, the material



Fig. 41 Belt-type magnetorheological polishing technology [188] a principle of belt-type magnetorheological polishing technology, b material removal function



Fig. 42 Surface morphology of SiC before and after polishing [189] a before polishing, b after polishing

removal rate of the workpiece per unit time increases. This technology adds an MPF circulation system, which guarantees the stability of the polishing process to a certain extent. The polishing equipment is integrated into the gantry machine, which further increases the application range of this technology.

However, the technology has a single polishing direction, and obvious polishing lines will remain on the surface of the polished workpiece, which limits the further improvement of the surface quality after polishing. Therefore, there are few researches on this technology, which are limited to the research content described in the above literature.

# 5.5.3 Reciprocating-type magnetorheological polishing technology

The processing feature of the reciprocating-type magnetorheological polishing technology is that the magnetic field generating device can move reciprocally during the processing process, which effectively solves the problem of single processing texture caused by other processing methods. Its processing principle is shown in Fig. 43. The workpiece is held by a fixture mounted on the spindle. The liquid carrier tank contains the magnetorheological polishing liquid, and a magnetic field-generating device



Rack 2. Workbench 3. Magnetic field generating device
 The polishing groove 5. Fluid 6. Workpiece 7. Fixture
 Machine tool spindle

Fig. 43 Principle of reciprocating-type magnetorheological polishing technology [190]

is placed below it. In the polishing process, the rotation of the spindle drives the workpiece to run at high speed, and the reciprocating mechanism drives the reciprocating movement of the magnetic field generator. The purpose of material removal is achieved through the mutual movement of the ribbon protrusion and the workpiece.

Li [190] conducted polishing experiments on K9 glass with self-designed equipment, which reduced the surface roughness of glass by 37% and made the material removal rate reach 10 nm/min. The feasibility of this technology is verified. Ma [191] used an orthogonal experiment to optimize the process parameters and processed K9 glass under the optimal process parameters. The surface roughness of K9 glass was reduced by 88%, and the material removal rate reached 14.1 nm/min. By comparing the processing performance of this technology before and after optimization, it can be seen that the material removal efficiency of this technology is low. As a result, the development of the technology stalled for several years. Until 2021, Wang et al. [192] continued their research on reciprocating-type magnetorheological polishing. Based on the previous study, the permanent magnets used in the magnetic field generation device were replaced with electromagnets, which improved the material removal rate of K9 glass by 67%, reaching 44.3 nm/min. Wang et al. [193] optimized the process parameters (working gap, processing time, etc.) for reciprocating-type magnetorheological polishing of K9 glass by single-factor experiments. It was concluded that a high-quality surface with a surface roughness of 28 nm was obtained with the optimal process parameters. Further, Wang et al. [194] adopted the response surface method to establish the prediction model of MRR and  $R_a$ . After experimental verification, the deviation between the actual value of MRR and the predicted value was 2.31%, and  $R_a$  was 5.12%. The accuracy of the prediction model is illustrated.

In summary, this technology has carried out relevant research for uniform texture and smoothness of the surface texture of the workpiece, which changes the surface quality of the processed workpiece to a certain extent. The equipment structure is simple and the cost is small.

However, this technology still has certain limitations. It is mainly reflected in the following aspects: after experimental studies, it was confirmed that the material removal rate of this technology was low. Although the processing quality of the workpiece is improved to a certain extent, there is still a big gap compared to other polishing technologies. Moreover, the processing materials of the existing studies are only for optical glass, and no other materials are involved. Therefore, the technology needs further development.

### 6 Comparison of several magnetorheological polishing technology and the proposal of curvature adaptive magnetorheological polishing technology

The processing accuracy of magnetorheological polishing technology not only changes with the need for scientific and technological reform, but also varies with the material, use and processing difficulty of the workpiece being processed in the same period. Therefore, MRR,  $R_a$ , surface uniformity and application range of the above typical magnetorheological polishing technologies were compared, as shown in Table 4.

As mentioned earlier, wheel-type magnetorheological polishing technology is widely used to finish aspheric and sphere surfaces. Because of its advantages, such as, mature equipment, good stability, and not being limited by the shape of the workpiece. However, its material removal rate is low. Cluster-magnetorheological polishing technology has the advantages of all-round, high-efficiency and high-quality processing. However, the MPF of this technology is difficult to circulate and renew, and complex surface processing is difficult. Ball-type magnetorheological polishing technology has unique advantages in processing shaped parts and small-diameter workpieces. However, its material removal efficiency is low and unsuitable for efficient and low-cost ultra-smooth surface polishing processing. Disc-type magnetorheological polishing technology effectively improves the processing efficiency of workpieces. However, the MPF is difficult to circulate and renew and unsuitable for complex surface polishing. The rotation-type overcomes the problems of low processing efficiency and poor processing quality of traditional machining technology. It is mainly used for the

Category	MRR	R <sub>a</sub>	Uniformity	Application
Wheel-type	Low	Smaller	Better	Aspheric surface and sphere, etc.
Cluster-type	Middle	Small	Good	Mainly planar
Ball-type	Low	Smaller	Good	Free-form surface, plane, etc.
Disc-type	High	Smaller	Better	Mainly planar
Rotation-type	High	Small	Better	Blind hole, cavity, etc.
Belt-type	High	Middle	Bad	Large aperture plane
Reciprocating-type	Middle	Middle	Bad	Mainly planar

 Table 4
 Comparison of several typical magnetorheological polishing technologies

machining of blind cavity parts. However, this technology has high requirements for detection technology and precision control technology in the machining process, and the technical level of the existing research is limited. The belttype increases the curvature radius of the polishing area and makes the contact area between the workpiece and the MPF increase. That is, the material removal rate of the workpiece per unit time increases. However, the polishing direction of this technology is single, and obvious polishing lines will remain on the surface of the polished workpiece, which limits the further improvement of the surface quality after polishing. The reciprocating-type has simple structure and small cost, and is mainly used for the processing of flat parts. However, its material removal rate is poor, and the MPF circulation is difficult.

In summary, "point contact" can be adapted to processing various face-type workpieces, but its material removal rate is low. "Face contact" is only applicable to the processing of flat workpieces. The MPF circulation renewal also plays a big role in the polishing effect. Therefore, the author proposed a curvature adaptive magnetorheological polishing technology with a circulatory system to achieve high efficiency and high-quality machining that can adapt to various surface shapes. This technology takes into account the idea of "point contact" and "surface contact". The control system can correct the position accuracy of the polishing head and workpiece according to different face shape requirements, thus improving the efficiency of workpiece processing. At the same time, the timely renewal of MPF also provides an important guarantee for the polishing effect. The polishing principle is shown in Fig. 44. Currently, the research in this area is still in the initial stage.

Magnetorheological polishing technology can provide an effective solution in areas requiring high-precision surface polishing. Suzhou Langxin Precision Optics Co., Ltd provides two MRP series of magnetorheological polishing application examples. Case 1, magnetorheological polishing equipment of the MRP series, was used for 25 h to polish a 422 mm calibre, coaxial paraboloid, microcrystalline glass. The RMS of the microcrystalline glass was made to



Fig. 44 Principle of curvature adaptive magnetorheological polishing technology with circulatory system

converge from 0.333  $2\lambda$  to 0.011 $\lambda$ . Case 2, magnetorheological polishing equipment of the MRP series, was used to polish a 200 mm calibre, secondary surface, fused silica for 12 h. The RMS of fused quartz converges from 0.308  $2\lambda$  to  $0.010 2\lambda$ . Li [25] of Changchun Institute of Optical Machinery also mentioned an example of magnetorheological polishing in his doctoral thesis. A self-developed wheel-type magnetorheological polishing equipment was used to polish a 1 500 mm calibre, off-axis aspherical, SiC for 40 h. The PV value of SiC was rapidly converged from 1 820.6 nm to 394.1 nm. The high efficiency and precision deterministic machining of large aperture aspherical magnetorheological polishing is realized. In addition, the United States QED company also provides magnetorheological polishing examples on the official website. Q series polishing equipment is used to polish the concave spherical surface with a large aperture and small curvature, and a high-quality machining effect is obtained. However, the application examples of magnetorheological polishing are limited, and subsequent development is still needed.

# 7 Development trend of magnetorheological polishing technology

Magnetorheological polishing technology has been an emerging ultra-precision machining method in recent years. It has attracted many scholars at home and abroad to study and made positive progress. In order to further promote the engineering application of magnetorheological polishing technology, the following prospects are proposed.

- (i) Diversity of processing materials. Currently, the most important applications of magnetorheological polishing technology are on non-magnetic materials such as optical glass, ceramics, and some alloys. Few studies have pointed out that magnetorheological polishing technology can be applied to process magnetic materials. Therefore, the application range of magnetorheological polishing technology in machining materials can be expanded.
- (ii) Focus on economic and environmental protection. The magnetorheological polishing mentioned above is processed using conventional abrasive grains. However, the manufacturing and processing of the abrasive itself cause energy and material consumption, and the configuration of the MPF containing chemical components can cause environmental pollution. In addition, the circulation of the MPF during the magnetorheological polishing process is one of the key factors affecting its cost. Therefore, future research on developing new green MPFs and optimizing the design of circulation devices can be carried out. This will also become an important research direction of magnetorheological polishing technology.
- (iii) Enrich the theoretical system. At present, the research theory on magnetorheological polishing still needs to be completed. It is still in the developmental stage for studying the mechanism of microscale processing of material nanoscale surfaces. Some things could still be improved in applying the existing mathematical material removal model by magnetorheological polishing to guide production. Therefore, a complete system can be established to explain the mechanism of nanoscale microscale processing of materials. A complete mathematical material removal model can be established to guide production, considering the actual working conditions. In addition, the future can continue to carry out thermal, optical, electrical, acoustic and chemical multi-field coupling processing. A new method of magnetorheological composite polishing was explored to provide a new theoretical basis for magnetorheological polishing.
- (iv) Intelligence. The control strategy and magnetorheological polishing method is also a hot topic in current

research. Researchers have always pursued intelligent equipment to reduce the dependence of human experience and the influence of interference from external factors on experimental results. Therefore, the intelligent degree of magnetorheological polishing equipment is directly related to the stability and efficiency of machining. At present, the intelligent degree of magnetorheological polishing equipment needs to be improved, and collaborative control among various systems needs to be developed. Therefore, the design of magnetorheological intelligent polishing equipment can be studied in the future.

(v) Online inspection. For the surface quality inspection of the workpiece after magnetorheological polishing, the most used is still offline inspection, which will affect the workpiece's processing efficiency and quality to a certain extent. The use of in-line inspection technology can avoid the above problems. However, at present, online inspection technology still needs to be improved. The main reasons are limited by various conditions, such as the difficulty of detecting sensor installation and magnetorheological fluid flow. Therefore, relevant research on the online detection of magnetorheological polishing technology can be carried out in the future. This will also be an inevitable trend.

Acknowledgements This work was supported by the National Natural Science Foundation of China (Grant No. U19A20104), the Natural Science Foundation of Jilin Province (Grant No. YDZJ202201ZYTS534), Jilin Provincial International Cooperation Key Laboratory for High-Performance Manufacturing and Testing (Grant No.20220502003GH).

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

### References

- 1. Dong K (2022) Interpretation of the "14th Five-Year Plan" intelligent manufacturing development plan and trend research. China Ind Inf Technol 1:24–29
- Xu H (2022) Vigorously develop high-end equipment manufacturing. China Electron News. https://doi.org/10.28065/n.cnki. ncdzb.2022.001097

 Huang KT, Gong H, Chen XB (2022) Study of ultra-precision turning path calculation of freeform surface with free contours. Int J Adv Manuf Technol 121:5451–5462

 Zhong YY, Dai YF, Xiao H et al (2021) Experimental study on surface integrity and subsurface damage of fused silica in ultraprecision grinding. Int J Adv Manuf Technol 115:4021–4033

- Zhang SJ, Zhou YP, Zhang HJ et al (2019) Advances in ultraprecision machining of micro-structured functional surfaces and their typical applications. Int J Mach Tool Manuf 142:16–41
- Zhu WL, Beaucamp A (2020) Compliant grinding and polishing: a review. Int J Mach Tool Manuf 158:103634. https://doi.org/10. 1016/j.ijmachtools.2020.103634
- Yang YK, Wang ZK, Pang MH et al (2022) The effect of process parameters on chemical mechanical polishing of quartz glass. Int J Mod Phys B 36:2240032. https://doi.org/10.1142/S021797922 240032X
- Yuan JL, Zhang FH, Dai YF et al (2010) Development research of science and technologies in ultra-precision machining field. Chin J Mech Eng 46(15):161–177
- Zhang J, Dai L, Wang F et al (2013) Restraint of mid-spatialfrequency error aspheric surface by small-tool adaptive polishing. Acta Opt Sin 33(8):202–208
- Xu J, Zou P, Kang D et al (2021) Theoretical and experimental study of bulge formation in laser polishing of 304 stainless steel. J Manuf Process 66:39–52
- 11. Peng YF, Yang L, Liu XY (2022) Polishing process and progress of calcium fluoride. Aeronaut Manuf Technol 65(13):14–25
- Seo J (2021) A review on chemical and mechanical phenomena at the wafer interface during chemical mechanical planarization. J Mater Res 36:235–257
- Subramani K, Vasudevan K, Karthik K et al (2022) Insights of abrasive water jet polishing process characteristics and its advancements. Mater Today Proc 52(3):1113–1120
- Xiao H, Dai Y, Duan J et al (2021) Material removal and surface evolution of single crystal silicon during ion beam polishing. Appl Surf Sci 544(1):148954. https://doi.org/10.1016/J. APSUSC.2021.148954
- Yin SH, Deng ZM, Guo YF et al (2020) Magnetorheological polishing using large polishing tool excited by electromagnetic field for silicon carbide wafer. Surf Technol 49(10):309–315
- Lv HF (2010) Research on removing subsurface damage by using magnetorheological finishing technology. Dissertation, National University of Defense Technology
- Li SY, Peng XQ (2013) Basic theory and method of controllable compliant tools for optic elements manufacturing. J Mech Eng 49(17):1–9
- Shi F, Dai YF, Peng XQ et al (2010) Removal of subsurface damage in grinding by magnetorheological finishing. Opt Precis Eng 18(1):162–168
- Lv HF, Zheng ZW, Peng XQ et al (2010) Research on removing subsurface damage by using magnetorheological finishing technology. Aviat Precis Manuf Technol 46(4):20–23
- Lu MM, Zhuang XL, Zhou JK et al (2021) Distribution model and optimization design of cluster magnetorheological field. Manuf Technol Mach Tool 10:45–49
- Rabinow J (1948) The magnetic fluid clutch. Trans Am Inst Electr Eng 67(2):1308–1315
- 22. Lu JY (2008) Research on inverted device for magnetorheological finishing. Dissertation, Harbin Institute of Technology
- Kordonski WI, Jacobs SD (1996) Magnetorheological finishing. Int J Mod Phys B 10:2837–2848
- Harris DC (2011) History of magnetorheological finishing. Proc SPIE 8016(4):561–566
- 25. Li LX (2016) Study on the key techniques of magnetorheological finishing for large aspheric optics. Dissertation, University of Chinese Academy of Sciences

- Li Z (2016) Research on the key technology of combined magnetorheological finishing. Dissertation, Hunan University
- Golini D, Kordonski WI, Dumas P et al (1999) Magnetorheological finishing (MRF) in commercial precision optics manufacturing. Proc SPIE 3782:80–91
- Lei LG (2021) Research on magnetorheological polishing technology of pseudo-random trajectory based on parametric grid. Dissertation, Jilin University
- 29. Li Y (2019) Study on the wear mechanism and suppression method of magnetorheological polishing wheel. Dissertation, China Academy of Engineering Physics
- QED Technologies Inc (2012) QED Championship date. QED Technologies, Rochester. https://qedmrf.com/
- Zhang F, Yu JC, Zhang XJ et al (1999) Magnetorheological finishing technology. Opt Precis Eng 7(5):1–8
- Zhang F, Zhang XJ, Yu JC et al (2000) Foundation of mathematics model of magnetorheological finishing. Opt Technol 26(2):190–192
- Zhang F, Pan SF, Zhang XJ et al (2004) Research on material removal of magnetorheological finishing. Opt Technol 27(6):522–523
- Zhang Y, Feng ZJ, Zhao GM (2004) Magnetorheological finishing tool and removal function. J Tsinghua Univ 44(2):190–193
- 35. Peng XQ, Dai YF, Li SY (2004) Material removal model of magnetorheological finishing. J Mech Eng 40(4):67–70
- You WW, Peng XQ, Dai YF (2004) MR fluids for finishing use. Opt Precis Eng 12(3):330–334
- Dai YF, Shi F, Peng XQ et al (2010) Deterministic figuring in optical machining by magnetorheological finishing. Acta Opt Sin 30(1):198–205
- Beier M, Scheiding S, Gebhardt A et al (2013) Fabrication of high precision metallic freeform mirrors with magnetorheological finishing (MRF). Proc SPIE 8884:139–152
- Schinhaerl M, Rascher R, Stamp R et al (2007) Filter algorithm for influence functions in the computer controlled polishing of high-quality optical lenses. Int J Mach Tool Manuf 47(1):107–111
- Kim DW, Cho MW, Lee JW et al (2009) Ultra-precision polishing of Si3N4 ceramics using magnetorheological fluids and diamond abrasives. J Ceram Process Res 10(3):290–295
- Huang W, Zhang YF, Zheng YC et al (2014) Magnetorheological finishing machine and technique based on two-polishinghead. J Appl Opt 35(3):494–499
- 42. Yan QS, Tang AJ, Lu JB et al (2008) Development of a new plate polishing technique with an instantaneous tiny-grinding wheel cluster based on magnetorheological effect. Adv Mater Res 53/54:155–160
- 43. Singh AK, Jha S, Pandey PM (2011) Design and development of nanofinishing process for 3D surfaces using ball end MR finishing tool. Int J Mach Tool Manuf 51(2):142–151
- 44. Singh AK, Jha S, Pandey PM (2012) Nano-finishing of a typical 3D ferromagnetic workpiece using ball end magnetorheological finishing process. Int J Mach Tool Manuf 63(4):21–31
- Niranjan M, Jha S, Kotnala RK (2014) Ball end magnetorheological finishing using bidisperse magnetorheological polishing fluid. Mater Manuf Process 29(4):487–492
- 46. Su YR, Li D, Liu HN et al (2013) Experiment research of polishing capability of magnetorheological finishing with a small permanent magnet ball-end tool. Key Eng Mater 589/590:497–501
- 47. Wang YQ, Yin SH, Huang H et al (2015) Magnetorheological polishing using a permanent magnetic yoke with straight air gap for ultra-smooth surface planarization. Precis Eng 40:309–317
- Li T, Xie MJ, Du WZ (2021) The necessity and development direction of ultra-precision machining are briefly discussed. Ind Heat 50(8):42–44

- Liu HN, Qin B, Wang YZ et al (2022) Interference analysis and ultra-precision grinding technology of hemispherical resonator curved surface machining. Diam Abras Eng 42(1):10–17
- Wang B, Zhang WL, Shi F et al (2022) Study on the repair technology of laser damage-fused silica optics based on the neural network method. Materials 15(15):5274. https://doi.org/10.3390/ MA15155274
- Zhang WY (2020) Research status and application progress of biomedical metal materials. Met World 2020(1):21–27
- Kunal A, Anant KS (2020) Magnetorheological finishing of UHMWPE acetabular cup surface and its performance analysis. Mater Manuf Process 35(14):1631–1649
- Yan QS, Zhao Y, Liang ZB et al (2021) Study on cluster magnetorheological finishing of medical titanium alloy with dynamic magnetic fields. Surf Technol 50(09):322–332, 378
- 54. Sun BY, Yu BJ, Tan HQ et al (2022) Magnetorheological finishing process of CoCrMo alloy. Surf Technol 51(10):310–320
- Chen BS, Zheng C, Huang DC (2021) Experimental study on magnetorheological polishing process of 316L stainless steel formed by SLM. Tool Eng 55(10):82–86
- 56. Paswan SK, Aggarwal A, Singh AK (2022) Investigation of magnetorheological fluid-based finishing process for improving the functional performance of hydraulic cylinder. Proc Inst Mech Eng E J Process 236(3):824–837
- 57. Singh M, Singh AK (2021) Magnetorheological finishing of copper cylindrical roller for its improved performance in printing machine. Proc Inst Mech Eng E J Process 235(1):103–115
- Tang YK (2017) Research on wear characteristic of fixed abrasive pad. Dissertation, Nanjing University of Aeronautics and Astronautics
- Zhang P, Yang JF, Qiu HD (2021) Material removal rate of double-faced mechanical polishing of 4H-SiC substrate. Int J Adv Manuf Technol 118:3989–3993
- Yang C, Zhang NW, Bai Y (2022) High-efficiency and high-quality combined polishing method of zinc selenide crystal. Infrared Laser Eng 51(9):37–42
- Peng XQ, Jiao FF, Chen HF et al (2011) Novel magnetorheological figuring of KDP crystal. Chin Opt Lett 9(10):102201. https:// doi.org/10.3788/COL201109.102201
- 62. Pan JS, Yan QS, Lu JB et al (2014) Cluster magnetorheological effect plane polishing technology. J Mech Eng 50(1):205–212
- 63. Mahender KG, Dinakara D, Inder MC et al (2021) Experimental investigation and machine parameter optimization for nano finishing of fused silica using magnetorheological finishing process. Optik 226:165908. https://doi.org/10.1016/j.ijleo.2020.165908
- 64. Pan JS, Zheng K, Yan QS et al (2020) Optimization study on magnetorheological fluid components and process parameters of cluster magnetorheological finishing with dynamic magnetic field for sapphire substrates. Smart Mater Struct 29(11):114009. https://doi.org/10.1088/1361-665X/abb988
- Jian VK, Ranjan P, Suri VK et al (2010) Chemo-mechanical magnetorheological finishing (CMMRF) of silicon for microelectronics applications. CIRP Ann Manuf Technol 59(1):323–328
- 66. Sun SL, Lu JB, Yan QS (2021) Technological experiment of cluster magnetorheological finishing of InP wafer with dynamic magnetic fields. Lubr Eng 46(11):40–45
- Chen XJ (2021) Research on plasma-assisted cluster magnetorheological finishing of single crystal SiC wafe. Dissertation, Guangdong University of Technology
- Guo MJ, Luo H, Wang CB et al (2018) Experimental study on magnetorheological finishing using large polishing tool for zirconia ceramic plane. Surf Technol 4(7):28–34
- 69. Xiao XL, Li GX, Mei HJ et al (2020) Polishing of silicon nitride ceramic balls by clustered magnetorheological finish. Micromachines 11(3):304. https://doi.org/10.3390/mi11030304

- Huang YP (2015) The development prospect of modern optical processing technology and industry. China High Tech Enterp 28:1–2
- Li XZ, Xu JK, Tong L et al (2021) Study on the influence of amplitude on ultrasonic assisted grinding of hard and brittle materials. Mater Sci Forum 1047:57–61
- 72. You K, Yan G, Luo X et al (2020) Advances in laser assisted machining of hard and brittle materials. J Manuf Process 58:677–692
- Xiong H, Luo YP, Wang WC et al (2019) Effect of surfactants with different HLB values on settling stability of magnetorheological fluid. Funct Mater 50(12):12126–12131
- Shu LF, Ni L, Xiang BP et al (2021) Effect of nonionic surfactant on the stability of water-based magnetorheological polishing. J Magn Mater Devices 52(5):73–78
- 75. Yang J, Yan H, Hu Z et al (2016) Viscosity and sedimentation behaviors of the magnetorheological suspensions with oleic acid/ dimer acid as surfactants. J Magn Magn Mater 417:214–221
- Niu FH, Hu ZD, Yan H et al (2018) Rheological properties of magnetorheological suspensions with PMMA/oleic acid as surfactants. Chem Ind Eng Prog 37(5):1888–1895
- Song KH, Park BJ, Choi HJ (2009) Effect of magnetic nanoparticle additive on characteristics of magnetorheological fluid. IEEE Ttans Magn 45(10):4045–4048
- Piao SH, Chae HS, Choi HJ (2015) Carbonyl iron suspension with core-shell structured Fe<sub>3</sub>O<sub>4</sub>@ SiO<sub>2</sub> nanoparticle additives and its magnetorheological property. IEEE Trans Magn 51(11):1–4
- 79. Guo QY, Wang J, Ouyang Q (2018) Effect of  $SiO_2$  with different specific surface areas on rheological properties and sedimentation stability of magnetorheological fluid. Mater Mech Eng 42(10):8–12
- Xiao Q, Wang JQ, Jin LP (2022) Research progress of key technology and process of magnetorheological polishing. Mater Rep 36(07):65–74
- Zhu XL, Sun HM, Liu NN et al (2019) Effect of magnetorheological fluids with different volume fractions on shear properties. Sci Technol Eng 19(14):183–186
- Xiao LJ, Wang CP, Wei J et al (2017) Experimental analysis of shear properties of magnetorheological fluids under magnetic fields. J Mater Sci Eng 35(4):659–622, 688
- Ginder JM, Davis LC (1994) Shear stress in magnetorheological fluids: role of magnetic saturation. Appl Phys Lett 65(26):3410–3412
- Zhao PH, Li XG, Tong Y et al (2021) Effect of the interface between magnetic particles and carrier liquids on magnetorheological properties and sedimentation of magnetorheological fluids: a molecular dynamics simulation and experimental insights. J Mol Liq 342:117377. https://doi.org/10.1016/j.molliq.2021. 117377
- Zhu YS, Gong XL, Li H et al (2006) Numerical analysis on shear yield stress of magnetorheological fluids. J China Univ Min Technol 35(7):498–503
- Li LX (2022) Magnetorheological effect and sedimentation stability of ionic liquid-modified silicone oil-based magnetorheological fluid. Dissertation, Dalian University of Technology
- 87. Bai Y (2015) Preparation of MR polishing fluid and study on the stability of removal function. Dissertation, University of Chinese Academy of Sciences
- Bai Y, Zhang F, Deng WJ et al (2014) Preparation of magnetorheological polishing fluid and its polishing stability. Acta Opt Sin 34(4):185–192
- Markus S, Elmar P, Andreas G et al (2005) Comparison of different magnetorheological polishing fluids. SPIE 5965:596528. https://doi.org/10.1117/12.656430

- Li LX, Liu ZY, Xue DL et al (2018) Rapid fabrication of a lightweight 2 m reaction-bonded SiC aspherical mirror. Results Phys 10:903–912
- Li LX, Zheng LG, Deng WJ et al (2015) Optimized dwell time algorithm in magnetorheological finishing. Int J Adv Manuf Technol 81(5):833–841
- 92. Shi F, Dai YF, Peng XQ et al (2009) Magnetorheological finishing for high-precision optical surface. Opt Precis Eng 17(8):1859–1864
- Wang F, Feng ZJ, Cheng HB (2006) Renewable magnetorheological finishing tool combined self-rotation and co-rotation movement. J Tsinghua Univ (Sci Technol) 2:184–186
- 94. Shu LF (2022) Design and performance study of arm type magnetorheological polishing machine and magnetorheological polishing fluid. Dissertation, Southwest University of Science and Technology
- Miao CL, Shaffir SN, Lambropoulos JC et al (2009) Shear stress in magnetorheological finishing for glasses. Appl Opt 48(13):2585–2594
- Miao CL, Lambropoulos JC, Jacobs SD (2010) Process parameter effects on material removal in magnetorheological finishing of borosilicate glass. Appl Opt 49(10):1951–1963
- Sidpara A, Jain VK (2012) Theoretical analysis of forces in magnetorheological fluid based finishing process. Int J Mech Sci 56(1):50–59
- Liu J, Li X, Zhang Y et al (2020) Predicting the material removal rate (MRR) in surface magnetorheological finishing (MRF) based on the synergistic effect of pressure and shear stress. Appl Surf Sci 504:144492. https://doi.org/10.1016/j.apsusc.2019.144492
- Liu SW, Wang HX, Zhang QH et al (2020) Regionalized modeling approach of tool influence function in magnetorheological finishing process for aspherical optics. Optik 206:164368. https://doi.org/10. 1016/j.ijleo.2020.164368
- Bai Y, Zhang X, Yang C et al (2022) Material removal model of magnetorheological finishing based on dense granular flow theory. Light Adv Manuf 3:1–10
- 101. Schinhaerl M, Smith G, Stamp R et al (2008) Mathematical modelling of influence functions in computer-controlled polishing: Part II. Appl Math Model 32(12):2907–2924
- 102. Guo HR, Wu YB, Li YG et al (2012) Technical performance of zirconia-coated carbonyl-iron-particles based magnetic compound fluid slurry in ultrafine polishing of PMMA. Key Eng Mater 523/524:161–166
- 103. Chen H, Chen J, Wang B et al (2016) Normal contour error measurement on-machine and compensation method for polishing complex surface by MRF. In: The 8th international symposium on advanced optical manufacturing and testing technologies, Proc SPIE, vol 9683, pp 226–236
- 104. Schinhaerl M, Schneider F, Rascher R et al (2010) Relationship between influence function accuracy and polishing quality in magnetorheological finishing. In: The 5th international symposium on advanced optical manufacturing and testing technologies, Proc SPIE, vol 7655, pp 234–243
- 105. Xu C, Hu H, Peng XQ et al (2021) Optimization of magnetorheological finishing process for aluminum mirror with complex surface. Acta Aeronaut Astron Sin 42(10):267–276
- 106. Peng XQ, Dai YF, Li SY et al (2004) Dwell time algorithm for mrf of axis-symmetrical aspherical parts. J Natl Univ Defense Technol 3:89–92
- 107. Hu H, Peng XQ, Dai YF et al (2009) Algorithm and implementation of magnetorheological finishing with spiral scan mode. J Natl Univ Defense Technol 31(4):5–9
- Zhang YF, Wang Y, Wang YJ et al (2010) Dwell time algorithm based on optimization theory for magnetorheological finishing. J Appl Opt 31(4):657–662
- 109. Li LX, Zheng LG, Deng WJ et al (2015) Magnetorheological finishing for curve surface based on 4-axis machine. Opt Precis Eng 23(10):2819–2826

- 110. Qian DB, Jiang C, Yao L et al (2019) The error analysis and simulation of magnetic compound fluid polishing for aspheric components. Opt Instrum 41(5):59–64
- Huang T, Zhao D, Cao Z (2020) Trajectory planning of optical polishing based on optimized implementation of dwell time. Precis Eng 62:223–231
- 112. Gao B, Wang J, Fan B et al (2023) Particle swarm optimization algorithm for dwell time of optical element in magnetorheological finishing. Semicond Optoelectron 44(3):395–399
- Li LX, Deng WJ, Zhang BZ et al (2014) Dwell time algorithm for large aperture optical element in magnetorheological finishing. Acta Opt Sin 34(5):217–223
- Cheng RM, Li LX, Li XC et al (2022) High-precision magnetorheological finishing based on robot by measuring and compensating trajectory error. Opt Express 30(25):44741–44768
- 115. Song C, Dai YF, Peng XQ et al (2010) Post processing for magnetorheological finishing of optical mirrors. Opt Precis Eng 18(8):1715–1721
- Song C, Dai Y, Peng X (2010) Model and algorithm based on accurate realization of dwell time in magnetorheological finishing. Appl Opt 49(19):3676–3683
- Dong ZC, Cheng HB, Tam HY (2014) Modified dwell time optimization model and its applications in subaperture polishing. Appl Opt 53(15):3213–3224
- Byung CK, Jae HC, Myeong WC et al (2018) Magnetorheological fluid polishing using an electromagnet with staight polepiece for improving material removal rate. J Mech Sci Technol 32(7):3345–3350
- Liu ZJ, Li ZY, Wang Z et al (2008) Research on the subsurface damage during finishing for optics. Aviat Precis Manuf Technol 44(5):1– 3, 16
- 120. Shi F, Qiao S, Tian Y et al (2022) The generation mechanism and suppression method of the "comet tail" defect on the single-crystal silicon surface polishing with magnetorheological. Ceram Int 48(12):17621–17631
- Xu L, Wang L, Chen HY et al (2022) Research progress in ultra-precision polishing of tungsten and its alloys. Surf Technol 51(4):24–36
- Pan JS, Yan QS, Xu XP et al (2013) Cluster magnetorheological effect plane polishing on SiC single crystal slice. China Mech Eng 24(18):2495–2499
- 123. Pan JS, Guo ML, Yan QS et al (2019) Research on material removal model and processing parameters of cluster magnetorheological finishing with dynamic magnetic fields. Int J Adv Manuf Technol 100(9/12):2283–2297
- 124. Guo ML, Yan QS, Pan JS et al (2018) Mechanism and experimental study of cluster magnetorheological polishing with dynamic magnetic. Diam Abras Eng 38(1):89–93+97
- 125. Yan QS, Liao BT, Lu JB et al (2021) Experimental study on cluster magnetorheological variable gap dynamic pressure planarization finishing. J Mech Eng 57(19):230–238
- 126. Luo B, Yan QS, Pan JS et al (2020) Uniformity of cluster magnetorheological finishing with dynamic magnetic fields formed by multi-magnetic rotating poles based on the cluster principle. Int J Adv Manuf Technol 107(1/2):919–934
- 127. Luo B, Yan QS, Cai JF et al (2022) An ultra-smooth planarization method for controlling fluid behavior in cluster magnetorheological finishing based on computational fluid dynamics. Precis Eng 74:358–368
- 128. Lu MM, Zhuang XL, Zhou JK et al (2022) A novel reciprocating cluster magnetorheological polishing device: design and investigation of removal model. Proc Inst Mech Eng B J Eng 237(9):1339–1352
- 129. Wu ZC, Yan QS, Bai ZW et al (2011) Ultra-smooth polishing research based on the cluster magnetorheological effect. Key Eng Mater 487:283–288

- Pan JS, Yan QS (2015) Material removal mechanism of cluster magnetorheological effect in plane polishing. Int J Adv Manuf Technol 81:2017–2026
- 131. Pan JS, Yu P, Yan QS (2018) Experimental investigations on the polishing forces characteristics of dynamic magnetic field magnetorheological effect polishing pad. J Mech Eng 54(6):10–17
- 132. Yan QS, Cai ZH, Pan JS et al (2022) Process and mechanism of magnetorheological variable gap dynamic pressure planarization finishing. Diam Abras Eng 42(4):488–494
- 133. Luo B, Yan QS, Cai JF et al (2022) Theoretical and experimental research into a novel method of cluster magnetorheological finishing based on a circular array polishing disk. Int J Adv Manuf Technol 121(9/10):6535–6550
- 134. Luo B, Yan QS, Pan JS et al (2022) Simulation and experimental research on magnetorheological finishing under dynamic pressure with a gap-varying. J Manuf Process 82:265–276
- 135. Yan QS, Liang ZB, Pan JS et al (2022) Modeling optimization and uniformity of cluster magnetorheological polishing via magnetic trajectory intensity in machining surface. Surf Technol 51(12):243–254
- Zhuang XL, Lu MM, Zhou JK et al (2022) Improved magnetorheological finishing process with arc magnet for borosilicate glass. Mater Manuf Process 37(4):458–466
- 137. Zhang LM, Li WX, Zhou JK et al (2023) Removal modeling and experimental verification of magnetorheological polishing fused silica glass. Micromachines 14(1):54. https://doi.org/10. 3390/mi14010054
- 138. Yan QS, Tang AJ, Lu JB et al (2008) An experimental study of planarization technique with instantaneous tiny grinding wheel cluster based on magnetorheological effect. Diam Abras Eng 5:66–70
- 139. Bai ZW, Yan QS, Lu JB et al (2014) Research on abrasive accommodate-sinking effect of cluster MR-effect polishing pad. China Mech Eng 25(20):2705–2710
- Bai ZW, Yan QS, Lu JB et al (2014) Parametric investigation into accommodate-sinking effect of cluster magnetorheological effect pad. Int J Adv Manuf Technol 75(9/12):1447–1456
- 141. Pan JS, Yu P, Yan JS et al (2017) An experimental analysis of strontium titanate ceramic substrates polished by magnetorheological finishing with dynamic magnetic fields formed by rotating magnetic poles. Smart Mater Struct 26(5):1–14
- 142. Yan QS, Xu SP, Lu JB et al (2017) Study on chemical reaction parameters of CMP solution for single-crystal SiC. Mach Des Manuf 9:98–100
- 143. Liang HZ, Lu JB, Pa JS et al (2018) Material removal process of sing-crystal SiC in chemical-magnetorheological compound finishing. Int J Adv Manuf Technol 94(5/8):2939–2948
- 144. Sidpara A, Jain VK (2012) Nanofinishing of freeform surfaces of prosthetic knee joint implant. Proc Inst Mech Eng B J Eng 226(11):1833–1846
- 145. Yin SH, Xu ZQ, Chen FJ et al (2013) Inclined axis magnetorheological finishing technology for small aspherical surface. J Mech Eng 49(17):33–38
- 146. Xu ZQ, Yin SH, Chen FJ et al (2013) Combined process of ultra-precision turning and polishing technology for small aspheric surface. Nanotechnol Precis Eng 11(6):479–484
- 147. Yin SH, Gong S, He BW et al (2018) Development on synergistic process and machine tools integrated inclined axis grinding and magnetorheological polishing for small aspheric surface. J Mech Eng 54(21):205–211
- 148. Wang HJ, Zhang FH, Zhao H et al (2007) Effect of several processing parameters on material removal ratio in ultrasonicmagnetorheological compound finishing. Opt Precis Eng 15(10):1583–1588
- 149. Wang HJ, Zhang FH, Zhao H et al (2007) Study on optical polishing of optical glass by means of

ultrasonic-magnetorheological compound finishing. Proc SPIE Int Soc Opt Eng 6722:67221P. https://doi.org/10.1117/ 12.783019

- 150. Wang HJ (2007) Research on the key technologies of ultrasonic-magnetorheological compound finishing. Dissertation, Harbin Institute of Technology
- 151. Zafar A, Sunil J (2017) Modeling of surface roughness in ball end magnetorheological finishing (BEMRF) process. Wear 374/375:54–62
- 152. Xu J, Li D, Chen MJ et al (2015) Process research of magnetorheological finishing with a small permanent magnet ball-end tool. Nanotechnol Precis Eng 13(4):244–250
- 153. Chen MJ, Liu HN, Cheng J et al (2017) Model of the material removal function and an experimental study on a magnetorheological finishing process using a small ball-end permanentmagnet polishing head. Appl Opt 56(19):5573–5582
- 154. Wang HJ, Zhang FH, Luan DR (2007) Research on material removal of ultrasonic-magnetorheological compound finishing. Int J Mach Mater 2(1):50–58
- 155. Liu YM, Feng J, Gu Y et al (2020) Research on numerical simulation of composite materials vibration assisted polishing. Mech Eng 3:25–29
- 156. Singh AK, Jha S, Pandey PM (2015) Performance analysis of ball end magnetorheological finishing process with MR polishing fluid. Mater Manuf Process 30(12):1482–1489
- 157. Singh G, Singh AK, Garg P (2017) Development of magnetorheological finishing process for external cylindrical surfaces. Mater Manuf Process 32(5):581–588
- Kansal H, Singh AK, Grover V (2018) Magnetorheological nano-finishing of diamagnetic material using permanent magnets tool. Precis Eng 51:30–39
- Bedi TS, Singh AK (2018) Development of magnetorheological fluid-based process for finishing of ferromagnetic cylindrical workpiece. Mach Sci Technol 22(1):120–146
- Kumar S, Singh AK (2018) Nanofinishing of BK7 glass using a magnetorheological solid rotating core tool. Appl Opt 57(4):942–951
- 161. Sidpara A, Jain VK (2013) Analysis of forces on the freeform surface in magnetorheological fluid based finishing process. Int J Mach Tool Manuf 69:1–10
- 162. Tian JC, Chen MJ, Liu HN et al (2022) Study on mechanism of improving efciency of permanent-magnet small ball-end magnetorheological polishing by increasing magnet-orheological fuid temperature. Sci Rep 12:7705. https://doi.org/10.1038/ s41598-022-11937-8
- Singh H, Niranjan MS, Wattal R (2022) Experimental investigation into residual stress in ball end magnetorheological finishing. J Eng Res 10:82–94
- 164. Singh H, Niranjan MS, Wattal R (2020) A study for the nanofinishing of an EN-31 workpiece with pulse DC power supply using ball-end magnetorheological finishing. Stroj Vestn J Mech E 66(7/8):449–457
- 165. Mayank S, Pulak MP (2021) The influence of ultrasonic vibrations on material removal in the silicon wafer polishing using DDCAMRF: experimental investigations and process optimization. Proc Inst Mech Eng C J Mech 236(6):3198–3215
- 166. Hu JF (2018) Research on polishing zirconia ceramic with magnetorheological finishing-ultrasoniccomposite technology. Dissertation, Changchun University of Technology
- 167. Lqbal F, Alam Z, Khan DA et al (2022) Automated insular surface finishing by ball end magnetorheological finishing process. Mater Manuf Process 37(4):437–447
- 168. Yin SH, Wang YQ, Xu ZQ et al (2015) A surface contact magnetorheological plane polishing device and method with uniform magnetic field excitation. China Patent 103273385B

- 169. Yin SH, Wang YQ, Deng GJ et al (2013) Effects of permanent magnet excitation on material removal rate in area taking magnetorhelogical finishing. Adv Mater Res 797:401–404
- 170. Wu JZ, Yin SH, Yang SJ et al (2022) Study on magnetorheological nano-polishing using low-frequency alternating magnetic field. Adv Mech Eng 12(1):1–11
- 171. Hu H, Li SY, Guan F et al (2017) Disc type magnetorheological polishing device. China Patent 201710085353.X
- 172. Guan F, Hu H, Li SY et al (2018) A novel lap-MRF method for large aperture mirrors. Int J Adv Manuf Technol 95(9/12):1–13
- 173. Wang YQ, Yin SH, Li YP et al (2017) Effects of translational movement on surface planarity in magnetorheological planarization process. J Mech Eng 53(1):206–214
- 174. Guan F, Hu H, Li SY et al (2018) Analysis of material removal rate and stability in lap-magnetorheological finishing. Opt Eng 57(5):055107. https://doi.org/10.1117/1.OE.57.5.055107
- 175. Luo H (2019) Study on damage generation mechanism and process of planar magnetorheological finishing using large polishing tool. Dissertation, Hunan University
- 176. Wang YQ (2016) Study on Magnetorheological finishing using large polishing tool for ultra-smooth flat surface. Dissertation, Hunan University
- 177. Wang YQ, Yin SH, Huang H (2016) Polishing characteristics and mechanism in magnetorheological planarization using a permanent magnetic yoke with translational movement. Precis Eng 43:93–104
- 178. Yang ZQ, Li H, Guo ZD (2019) Study on magnetorheological chemico-mechanical polishing technique for sapphire. J Xi'an Technol Univ 39(3):266–272
- 179. Yin SH, Wang YQ, Li YP et al (2016) Experimental study on magnetorheological chemical polishing for sapphire substrate. J Mech Eng 52(5):80–87
- 180. Sirwal SA, Singh AK (2018) Analysis of the surface roughness for novel magnetorheological finishing of a typical blind hole workpiece. Proc Inst Mech Eng C J Mech 233(5):1541–1561
- 181. Talwinder SB, Anant KS (2019) An initial new approach for magnetorheological finishing of ferromagnetic internal cylindrical surfaces. Int J Adv Manuf Technol 100(5/8):1017–1030
- 182. Aggarwal A, Singh AK (2021) Development of grinding wheel type magnetorheological finishing process for blind hole surfaces. Mater Manuf Process 36(4):457–478
- 183. Aggarwal A, Singh AK (2021) Experimental investigation for fine finishing of the tapered mould cavity using a newly developed GWMRF process. J Braz Soc Mech Sci 43(9):444. https:// doi.org/10.1007/s40430-021-03159-2
- 184. Sirwal SA, Singh AK (2019) Analysis of the surface roughness for novel magnetorheological finishing of a typical blind hole workpiece. Proc Inst Mech Eng C J Mech 233(5):1541–1561
- 185. Sunil KP, Auant KS (2021) Analysis of finishing performance in rotating magnetorheological honing process with the effect of particles motion. Proc Inst Mech Eng E J Process 235(4):1104–1118
- 186. Yahya C, Mohammad RR, Mohsen K et al (2022) Design and development of ultrasonic assisted-rotational magnetorheological abrasive flow finishing (UA-RMRAFF) process. Appl Acoust 197:108950. https://doi.org/10.1016/j.apacoust.2022. 108950
- 187. Luo X, Ren K, Hu HF et al (2014) Magnetorheological polishing device for ultra large diameter optical processing. China Patent CN201410120264.0
- Ren K, Luo X, Zheng LG et al (2014) Belt-MRF for large aperture mirrors. Opt Express 22(16):19262–19276
- 189. Wang DK (2018) Study on the key technologies of belt magnetorheological finishing. Dissertation, University of Chinese Academy of Sciences

- 190. Li SQ (2015) Study on the reciprocating magnetorheological polishing experimental machine and preparation and property of polishing fluid. Dissertation, Northeastern University
- 191. Ma HX (2017) Study on particle movement trajectories and process parameters optimization for reciprocating magnetorheological polishing. Dissertation, Northeastern University
- 192. Wang RS, Xiu SC, Sun C et al (2021) Study on material removal model by reciprocating magnetorheological polishing. Micromachines 12(4):413. https://doi.org/10.3390/mi120 40413
- 193. Wang RS, Sun C, Xiu SC et al (2022) Experimental study on reciprocating magnetorheological polishing. Ind Lubr Tribol 74(2):164–170
- 194. Wang RS, Sun C, Xiu SC (2022) Study of machining parameters in reciprocating magnetorheological polishing process based on response surface methodology. Ind Lubr Tribol 74(9):1007–1014



Ming-Ming Lu received a doctorate degree from Jilin University in 2014. He is now a master tutor and doctoral tutor at the School of Mechatronic Engineering, Changchun University of Technology. His research interests include micro-nano and ultraprecision manufacturing technology, vibration-assisted processing technology, etc.





**Ya-Kun Yang** was born in Kaifeng, Henan province, China in 1996. He is currently a doctoral student majoring in mechanical engineering at Changchun University of Technology. His main research includes but is not limited to precision machining and inspection technology, difficult-to-machine material processing, Magnetorheological polishing technology.

Jie-Qiong Lin was born in Changchun, Jilin Province. She is now the vice president of Changchun University of Technology, graduate supervisor and doctoral supervisor. Received a doctorate degree from Jilin University in 2005, mainly engaged in micro-nano and ultra-precision manufacturing, large-area laser processing, precision polishing and error compensation.



Yong-Sheng Du was born in Jilin, Jilin province, China in 1996. He received the B.S. degree in Mechanical Engineering from Changchun University of Technology, China, in 2015. He is currently a doctoral student majoring in mechanical engineering at Changchun University of Technology. His main research includes but is not limited to precision machining and inspection technology, difficultto-machine material processing, composite material processing and model simulation.



Xiao-Qin Zhou is now a master tutor and doctoral tutor at the School of Mechanical and Aerospace Engineering, Jilin University. His research interests include micro-nano and ultraprecision manufacturing technology, vibration-assisted processing technology, etc.